

# THE UNIVERSITY of EDINBURGH

# Edinburgh Research Explorer

# Transition versus physical climate risk pricing in European financial markets

A text-based approach

Citation for published version:

Bua, G, Kapp, D, Ramella, F & Rognone, L 2024, 'Transition versus physical climate risk pricing in European financial markets: A text-based approach', *The European Journal of Finance*, pp. 1-36. https://doi.org/10.1080/1351847X.2024.2355103

Digital Object Identifier (DOI): 10.1080/1351847X.2024.2355103

Link:

Link to publication record in Edinburgh Research Explorer

**Document Version:** Peer reviewed version

Published In: The European Journal of Finance

#### General rights

Copyright for the publications made accessible via the Edinburgh Research Explorer is retained by the author(s) and / or other copyright owners and it is a condition of accessing these publications that users recognise and abide by the legal requirements associated with these rights.

#### Take down policy

The University of Edinburgh has made every reasonable effort to ensure that Edinburgh Research Explorer content complies with UK legislation. If you believe that the public display of this file breaches copyright please contact openaccess@ed.ac.uk providing details, and we will remove access to the work immediately and investigate your claim.



# **Transition Versus Physical Climate Risk Pricing in European Financial Markets: A Text-Based Approach**<sup>\*</sup>

Giovanna Bua<sup>a</sup>, Daniel Kapp<sup>a</sup>, Federico Ramella<sup>a</sup>, Lavinia Rognone<sup>c,b,a,\*</sup>

<sup>a</sup>European Central Bank, 60314, Frankfurt am Main, DE <sup>b</sup>Alliance Manchester Business School, The University of Manchester, M15 6PB, UK <sup>c</sup>University of Edinburgh Business School, The University of Edinburgh, Edinburgh EH8 9JS, UK

#### Abstract

Under its climate regulation, the EU is expected to become the first continent with a net-zero emissions balance. We study the pricing of climate risks, physical and transition, within European markets. Using text-analysis, we construct two novel (daily) physical and transition risk indicators for the period 2005-2021 and two global climate risk vocabularies. Applying our climate risk indices to an asset pricing test framework, we document the emergence of economically significant transition and physical risk premia post-2015. From a firm-level analysis, using firms' GHG emissions, GHG emissions intensity, environmental, and ESG scores, we find that rises in transition (physical) risk are typically associated with an increase (decrease) in the return of green (brown) stocks. Firm-level information is used by investors to proxy firms' climate-risks exposure, especially for transition risk since 2015, whereas the sectoral classification appears to proxy firms' exposures to physical risk. From a country-level analysis emerges an intensified connection between European stock markets and climate risks post-2015, yet with some heterogeneity. Our results have important economic implications and show that investors demand compensation for their exposure to both climate risk types. Our novel climate risk vocabularies and indicators find several applications in identifying, measuring, and studying climate risks.

Keywords: Climate risks premia, Transition risk, Physical risk, ESG, Text-analysis

JEL Codes: C58, G12, G14, G28, Q51, Q54

<sup>\*</sup>The authors are grateful for helpful comments from Wolfgang Lemke, Isabel Vansteenkiste, Fabio Fornari, Stuart Hyde, S. Sarah Zhang, Andrew Urquhart, Yoichi Otsubo, Chi-Yang Tsou, Oguzhan Cepni, Harrison G. Hong, Carlos Cañizares, and comments and suggestions from participants at various conferences and seminars, including the European Central Bank, the Central Bank of Slovakia, and the Central Bank of Ireland seminar series, the University of Manchester, the 1st Conference on Sustainable Banking & Finance (CSBF) 2023, the Central Bank Research Association (CEBRA) Annual Meeting 2022, the 1st Conference on INternational Finance; Sustainable and Climate Finance and Growth (CINSC) 2022, the International Finance and Banking Society (IFABS) 2021 Oxford Conference, and the OECD's Committee on Financial Markets (2021). This research did not receive any specific grant from funding agencies in the public, commercial, or not-for-profit sectors. The authors report there are no competing interests to declare. Disclaimer: The views expressed in this paper are those of the authors and do not necessarily reflect the views of the European Central Bank. Any errors and omissions are the sole responsibility of the authors.

The climate risk indicators and vocabularies constructed in this paper can be downloaded from the Lavinia Rognone's library or from the Economic Policy Uncertainty website.

<sup>\*</sup>Corresponding author: Lavinia Rognone. Email address: lrognone@ed.ac.uk. Orcid: 0000-0002-0319-8812. University of Edinburgh Business School, The University of Edinburgh, 29 Buccleuch Place, Edinburgh EH8 9JS United Kingdom. Phone: +44 730 844 6733.

# 1. Introduction

As climate change progresses, investors are expected to reflect climate-related risks in firms' valuations. While this may seem obvious in light of the overarching evidence that climate change represents a source of financial risk, documenting climate risk premia is not as trivial, as demonstrated by conflicting results throughout the green finance literature (In et al. 2019; Hsu et al. 2023; Alessi et al. 2021; Bolton and Kacperczyk 2023; Pástor et al. 2022). Several factors might impede climate risk-informed investment decisions such as the lack of agreed metrics for firms' exposure to climate risks, alongside the difficulty of identifying and measuring climate risk events over time (Engle et al. 2020). It follows that investors might not be able to easily screen firms exposed to climate risks, failing to detect climate-risky investments (Bolton and Kacperczyk 2021). In contrast, market participants might be insensitive to climate change shocks, suggesting they do not perceive these as financial risks.<sup>1</sup> Both scenarios could lead to a mispricing of climate risks with consequences for the functioning of the financial sector as such and as a vehicle to transmit climate policies.

Besides the key role played by the EU in international environmental agreements such as the UN Climate Convention, the Kyoto Protocol, and the Paris Agreement, there is evidence of a "self-policy" that makes the EU progressing as an entity different from other countries when talking about climate regulation. In fact, the European Council has set the goal to reduce the EU greenhouse gas emissions by 55% by 2030, compared to 1990, and reach climate neutrality by 2050. In June 2021, the Council also adopted the "European climate law", part of the European Green Deal, under which EU countries are legally obliged to reach the 2030 and 2050 climate goals. To facilitate these objectives, the EU has released the "Fit for 55" package of proposals to revise and update the EU legislation, aims to spend 30% of the EU 2021-2027 budget to tackle

<sup>&</sup>lt;sup>1</sup>Unawareness and market inattention to the financial implications of climate risk can impact the pricing of such risks in assets. Engle et al. (2020) suggest that climate risk pricing may depend on investors' attention. Murfin and Spiegel (2020) and Hong et al. (2019) identify inefficiencies in climate risk pricing, partly attributing this to scepticism or market inattention to climate risks.

climate change, and has introduced a "Just Transition Mechanism" valued more than €100 billion and devoted to the green transition. Under these premises, the EU is expected to be the first continent to reach a net-zero emissions balance.

The EU adoption of a self-climate regulation and the ambitious roadmap to become the first net-zero emissions continent create room for research on understanding, exploring, and deepening the relationship between Europe, specifically, and climate change risks. With this paper, we therefore want to study whether climate risks, physical and transition, are priced in European markets documenting to which extent these are perceived as financial risks and whether investors consider some firms or activities as more exposed to climate risks than others. Nevertheless, to run a rigorous analysis of the different roles of physical and transition risk, it is pivotal to have as comprehensive, informative, and reliable measures of these as possible. Previous studies mainly capture sub-dimensions of physical and transition risk, e.g. natural disasters or climate policies, ignoring relevant aspects of these complex risks leading to restricted results. A deeper understanding of the impact of physical and transition risk as a whole remains largely unexplored and becomes essential to, e.g., make climate-informed investment decisions or make an effective climate risks management. Using text-analysis, we create two novel climate risks vocabularies which enjoy several advantages, improving on the existent literature and delivering sophisticated climate risk indicators allowing us to unveil new financial implications of climate change.

In particular, we distinguish between physical and transition risk using a text-based approach similar to Engle et al. (2020) acknowledging that the two can impact financial markets differently. Physical risk represents loss of value or costs due to chronic, e.g. sea level rise and drought, and/or acute, e.g. floods and heat waves, hazards. Transition risk arises from the costly adjustment towards a climate-neutral economy, typically prompted by climate policies, technological advances, and shifts in public preferences. To dissect climate change risks, we first examine scientific texts on climate change to build two novel *global* and comprehensive climate vocabularies on physical and transition risk. We then construct a Physical Risk Index (PRI) and Transition Risk Index (TRI) by comparing the vocabularies with a corpus of news sourced from Reuters News. This approach

posits that investors use news to update beliefs about climate change risks and that climate change news coverage intensifies with climate risks (Engle et al. 2020). The resulting PRI and TRI spike during days when the discussion on either risk type increases substantially. PRI captures multiple aspects of physical risk, e.g. detecting news concerning rising sea levels, permafrost thawing, floods, and adaptation measures, whereas TRI detects regulatory climate news as well as news discussing the importance of technological advances to achieve a greener economy, among others.

Drawing upon standard asset pricing theories, which suggest that assets hedging against future market conditions changes tend to have higher prices, implying lower expected returns (e.g. Breeden (1979)), and considering that climate change risks affect future investment and consumption opportunities (e.g. Huynh and Xia (2021)), it is expected that investors would prefer to hold assets that perform well in the face of increasing climate change risks, even if this entails accepting lower returns for such climate-hedging assets (e.g. Engle et al. (2020); Huynh and Xia (2021)).<sup>2</sup> We therefore employ the two novel physical and transition risk indices to gauge the presence of physical and transition climate risk premia in European equity markets. In particular, we adopt a standard portfolio sorting approach over the period from January 2005 to October 2021 and, in line with recent studies that document a higher importance of climate risks since the time of the Paris Agreement (e.g. Bolton and Kacperczyk 2023; Goldsmith-Pinkham et al. 2023), we further consider two sub-periods as before and after 2015 to investigate any time-varying relationship between equity markets and climate risks.

We perform time series regressions of equity returns on climate risks, controlling for the Fama and French (2015) five factors known to drive returns, and estimate equity market sensitivities to climate risks. The resulting loading on the climate risk factors, i.e. transition and physical risk betas, determine our estimates of firm-level climate risk exposure. Stocks with positive/high climate

<sup>&</sup>lt;sup>2</sup>Pástor et al. (2021) also predict that the lower expected return of green assets compared to brown assets reflects a risk premium, as greener assets offer better climate risk hedging. Additionally, under the carbon risk premium hypothesis of Bolton and Kacperczyk (2021), there exists a positive link between a firm's carbon emissions and its stock returns, indicating that investors demand compensation for holding stocks of high carbon emitters due to increased carbon risk. Pástor et al. (2021) suggest that price differentials between green and brown assets could also stem from green taste preferences.

risk beta perform well when climate risk increases offering potential hedge against climate change risks to investors who in turn should be willing to accept lower expected returns from such stocks. On the opposite, negative/low climate beta stocks perform badly when climate risks increase suggesting they are climate-risky stocks for which investors should require higher expected returns to compensate for the extra risk they'd bear if holding them within their portfolios. It follows that a low-minus-high (LMH) transition (physical) climate beta portfolio should therefore earn positive excess returns in case a climate risk premium exists. Results show the presence of economically significant transition and physical risk premia as sizable as 7.05% and 6.14% on average per year after 2015, respectively, providing a first-time empirical evidence of both climate-related risks pricing in European equity markets.<sup>3</sup> Results document that investors are demanding compensation for their exposure to physical and transition risk, and that such compensation increases in their exposure, as demonstrated by stronger results when grouping portfolios from quintiles to 25 percentiles.

Due to the lack of agreed metrics of firms' exposure to climate-related risks, we also run a firmlevel and sectoral-level analysis to investigate which type of information is more likely used by investors to detect physical and transition risky stocks, further contributing to Bolton and Kacperczyk (2021) who question whether investors consider the industry where firms operate in, rather than firm-level information, as material information on firms' exposure to climate risks. Compared to Bolton and Kacperczyk (2021), we consider a wider range of firm-level information, not only GHG emissions, alongside our comprehensive climate risk measures. We include PRI and TRI into a Fama and French (2015) five factors asset pricing model to test equity sensitivity to climate risks where: i) firms are sorted on GHG emissions levels, GHG emissions intensity, Environmental (E) scores, and Environmental, Social, and Governance (ESG) scores, with returns being aggregated into green and brown portfolios – firm-level analysis; and ii) firms are grouped

<sup>&</sup>lt;sup>3</sup>Using our comprehensive climate risk measures, we document the emergence of a transition risk premium in Europe, challenging prior findings that, relying solely on GHG emissions, identify Asia as entirely responsible for global transition risk premium around the Paris Agreement (Bolton and Kacperczyk 2023). Thus, we underscore the importance of precise climate risk measurement for uncovering additional economic implications.

by sector (NACE Rev. 2 classification) with returns aggregated accordingly – sectoral-analysis. Overall, firm-level information appears to be used as a gauge for firms' exposure to climate risks, especially for transition risk and since 2015. Rises in transition (physical) risk typically increase (decrease) the excess return of green (brown) stocks under the firm-level analysis. In contrast, the sectoral classification appears sufficient to identify firms' exposure to physical risk. Additionally and motivated by the climate economics literature that underscores the importance of considering geographical locations of climate policies and physical risk (Nordhaus and Yang 1996; Bolton and Kacperczyk 2023; Cruz and Rossi-Hansberg 2023), we provide some insights on European individual countries' sensitivity to climate-related risks over time as an additional test. Our findings suggest that a country-level relation between the stock market and climate risks exists and has increased post-2015, with countries overall experiencing more severe association with climate risks. Yet, the results show some heterogeneity country-by-country further underscoring the importance of the consideration of sub-national conditions in both investment decision-making and when designing climate policies.

This paper contributes and relates to a growing strand of literature which focuses on understanding the impact of climate risks on asset prices.<sup>4</sup> Climate-related risks pose relevant economic challenges (see e.g. Stern 2007; Pankratz et al. 2023), can cause stress in the financial system (Flori et al. 2021), drive the cross-section of both stock (e.g. Hsu et al. 2023; Bolton and Kacperczyk 2023) and bond returns (e.g. Krueger et al. 2020; Painter 2020; Huynh and Xia 2021),<sup>5</sup> and influence relative dynamics between green and brown equities (Bouri et al. 2022) or between ESG and conventional assets (Cepni et al. 2023). Climate-related risks also impact other asset classes,

<sup>&</sup>lt;sup>4</sup>For a more complete review see Hong et al. (2020) and Giglio et al. (2021).

<sup>&</sup>lt;sup>5</sup>Differently from Krueger et al. (2020) who consider survey-based data at the sectoral level to examine the implications of climate risks for equity valuations, we employ innovative climate risk metrics derived from objective sources like climate news shocks, as opposed to surveys, and conducts analyses at multiple levels, including sectoral, firm-level, and selected portfolio assessments. Furthermore, whereas Painter (2020) explores climate risk pricing in municipal bonds, motivated by examining whether investors price climate risks when these cannot be easily hedged or eradicated, we focus on the stock market and consider, for instance, sectors with varying degrees of expected climaterisk exposure (e.g. from mining and quarrying to arts), contributing to the initial question of Painter (2020). Finally, Painter (2020) focuses on a single physical risk, whereas we use comprehensive climate risks measures and separate physical and transition risks.

such as foreign exchange markets (Bonato et al. 2022), gold (Cepni et al. 2022; Salisu et al. 2023), real estate (e.g. Bernstein et al. 2019; Baldauf et al. 2020; Murfin and Spiegel 2020), and financial institutions more broadly (e.g. Battiston et al. 2021; Giglio et al. 2021; Do et al. 2022). In addition to the aforementioned literature, theoretical underpinnings from other scientific research, including publications by the IPCC, underscore the pressing risks associated with climate change and emphasize the pivotal role that the financial sector and the broader economy can play in mitigating these risks (IPCC 2023). On one hand, the alarming consequences of climate change and their connections to financial markets motivate our research question to identify an actual climate risk premium within the stock market. On the other hand, there are profound implications of mispricing climate change risks that necessitate assessment. Climate risk mispricing can, in fact, have detrimental effects on the economy through various channels, including financial stability concerns, the emergence of stranded assets, a more abrupt response to future green transitions, the financial sector's inability to facilitate mitigation policies, and more (Drudi et al. 2021; ECB 2023).

The equilibrium model of Pástor et al. (2021) predicts that green assets have lower expected returns compared to brown assets, but they have higher realized returns when agents are surprised by climate change concerns. While this conjecture appears convincing, empirical evidence on the presence of carbon risk premia is not yet conclusive. Some studies find that investors require additional compensation for holding brown assets, especially after the Paris Agreement, whereas others provide no evidence of price differentials (see In et al. 2019; Hsu et al. 2023; Alessi et al. 2021; Bolton and Kacperczyk 2023; Pástor et al. 2022). Additionally, transition risk does not solely represent a carbon/GHG risk, but it also stems from public shifts in preferences toward greener products, technological advances, and changes in environmental regulation – all aspects that jointly determine the financial risks posed by transition risk. For instance, Adomako et al. (2023) show that stakeholder green pressures, e.g. demand for green products, determine an increase of ecoproduct innovation and Dechezleprêtre et al. (2021) find higher valuations to firms that engage in clean research and development activities. Sena et al. (2022) find that the reshoring of subsidiaries, associated with relocation costs, is more likely in countries with environmental disclosure regula-

tions. Therefore, the green finance literature highlights the need for studying transition risk using enhanced and more comprehensive risk metrics. Similarly, the literature on the financial consequences of physical risk mainly considers specific risk events, such as sea level rise (Painter 2020; Goldsmith-Pinkham et al. 2023), high temperatures (Addoum et al. 2020), droughts (Hong et al. 2019), hurricanes (Kruttli et al. 2021), or other generic extreme events (Baltas et al. 2022), leaving ample room for analyses on the financial implications of physical risk using more complete risk measures.<sup>6</sup> Goldsmith-Pinkham et al. (2023) ignore.

As contributions to these studies, we: i) consider both aspects of climate risks and build comprehensive measures of physical and transition risks which allow us to both document climate risk premia exhaustively and attribute which of the two risk types drives any increase/decrease in green/brown stocks returns, refining or overturning previous findings; ii) focus on Europe, offering an informative study for, but not limited to, European institutions and investors; iii) run a daily analysis, particularly relevant for short-horizon investors and regulators seeking timely decisions, and unlike studies that use lower frequency data or focus on long-run climate implications (e.g. Bansal et al. 2016; Engle et al. 2020); iv) consider both the full sample and sub-periods to expand further the existing literature on the time-varying effects of climate risk; and v) consider various indicators of climate risk exposure, documenting which ones may be used by investors.

This paper also relates to the strand of the climate finance literature which uses text analysis to measure climate risks (Batten et al. 2016; Ardia et al. 2023; Engle et al. 2020; Meinerding et al. 2023; Faccini et al. 2023). While previous studies improve upon climate risk identification, they either consider climate change as a unique risk factor (Engle et al. 2020), focus only on transition risk (Batten et al. 2016; Meinerding et al. 2023), or focus on specific sub-categories of physical and transition risks (Ardia et al. 2023; Faccini et al. 2023).<sup>7</sup> Our paper improves the existent literature

<sup>&</sup>lt;sup>6</sup>Compared to Goldsmith-Pinkham et al. (2023), for instance, our analysis is not limited to the specific climaterelated risks of sea level rise and floods. Instead, our study encompasses different types of physical risks, which we integrate into a comprehensive index. This allows us to thoroughly explore whether there are economic and financial impacts arising from the various sources of physical climate risks. In addition, we further investigate the other pivotal component associated with climate change finance, namely the transition risk component, which

<sup>&</sup>lt;sup>7</sup>Using a textual analysis approach, Ardia et al. (2023) identify eight climate change sub-categories. Faccini et al. (2023) filter news by "climate change" and "global warming" and employ a Latent Dirichlet Allocation approach to

as our physical and transition risk vocabularies are global vocabularies, i.e. can be applied to any text source, that originate comprehensive climate risk indicators able to capture the entire multifaceted characteristics of the two climate risks without discarding relevant categories.

The remainder of this paper is organised as follows: Section 2 describes the text analysis methodology and provides a discussion of the physical and transition risk indices. Section 3 describes transition and physical risk pricing methodology. Section 4 provides an overview of the data used in this study. Section 5 lays out the main results. Section 6 concludes.

### 2. Measuring Climate Risk through Text-Analysis

To test financial markets sensitivity to physical and transition risks we need proxies to measure these risks. We exploit newspaper content to identify physical and transition risk shocks expanding upon the text-based approach of Engle et al. (2020) – who proxy innovations to climate change news, but without distinguishing between physical and transition risk. In particular, we compare authoritative texts on climate change with a corpus of European news sourced from Reuters News<sup>8</sup> based on the assumption that events covered in newspapers can carry relevant information on climate change. We then create two separate vocabularies and use these to construct distinct physical and transition risk shock indices. Recent studies have started relying on our measure of physical and transition risks for various analyses (see, for instance, Blasberg et al. (2022); Cepni et al. (2022); Pietsch and Salakhova (2022); Ali et al. (2023); Bouri et al. (2023a,b); Cepni et al. (2023); Goodell et al. (2023); Gupta and Pierdzioch (2023); Karmakar et al. (2023); Arfaoui et al. (2024); Fava et al. (2024)).

cluster news topics into "Natural Disasters", "Global Warming", "International Summit", and "U.S. Climate Policy" factors.

<sup>&</sup>lt;sup>8</sup>Reuters provides news to professionals via desktop terminals, world's media organizations, industry events, and consumers. Reuters News also includes the Breakingviews.com content (Source reuters.com and reutersagency.com, accessed on 16/06/2021).

#### 2.1. Physical and Transition Risk Vocabularies

To construct the climate risk vocabularies, we follow three main steps. First, we select a large number of scientific and authoritative texts on the topic of climate change published by governmental authorities and other institutions, starting with the collection already adopted by Engle et al. (2020). We screen texts' content and retain those whose content can be associated with either physical risk or transition risk topics. We further add financial texts describing both risk types as a genuine attempt to construct risk measures which incorporate multiple perspectives. The complete list of texts is summarised in Table A1 in the Appendix. We then aggregate the 13 (10) texts covering physical (transition) risk to create a single document on physical (transition) risk. The idea is that if one reads the whole physical risk document and transition risk. The two climate documents in fact, aggregating authoritative and scientific publications about climate change, represent our full information about climate risks which we in turn want to find within the news coverage to construct our climate risk indicators, and therefore represent the information we use to feed our text-based algorithms.

As a second step, we create two lists of unique stemmed unigrams and bigrams, jointly referred to as terms, with the associated term frequency scores (*tf*) from the physical and transition risk documents. Then, we create an analogous list of terms and frequencies from Reuters News, where real-time news are aggregated into daily documents. To do so, we retrieve a total of over 2.5 million real-time news in English language from the Factiva database over the period Jan 2005-Oct 2021.<sup>9</sup> Thereafter, we apply a one-day novelty filter to the sample to eliminate redundancy among the data. Specifically, only the first news of the day is kept from a series of similar news published on the same day (see Dang et al. (2015), Rognone et al. (2020), and Faccini et al. (2023)) and only news published during days in which European equity markets are open for trading are retained.<sup>10</sup>

<sup>&</sup>lt;sup>9</sup>The Factiva database allows users to select a regional focus of the news. We therefore consider Reuters News with a European regional focus (news related to Europe or its constituent countries and regions). It is important to note that, although these news have to be related to Europe, they may also discuss other countries or international issues.

<sup>&</sup>lt;sup>10</sup>News which corpus length exceeds 5,000 words are not included in our analysis for both computational reasons

The final sample counts 1,096,392 news.

Third, we convert the physical (transition) risk document and each daily news document into term frequency-inverse document frequency (tf-idf). Terms earn high tf-idf if they are representative of the individual text. This means that they are frequent within the document (high tf) and infrequent among other documents (high idf). On the opposite, low tf-idf score terms are common to many documents (low idf) and/or very infrequent within the document (low tf) and therefore have poor ability in representing the content of the individual text (Engle et al. 2020; Gentzkow et al. 2019). By multiplying the tf scores of the physical risk and transition risk documents by their relative idf scores from the collection of news,<sup>11</sup> we are able to obtain vocabularies ranked by term relevance (tf-idf).

Thanks to our methodology, which combines the *tf-idf* with the screening of texts on the topic of physical and transition risks, the resulting climate risk vocabularies benefit from several advantages, improving on the existent literature and delivering sophisticated climate risk indicators: i) the phraseology associated with the two types of risks is extracted from the authoritative texts rather than being defined ex-ante by the authors, *no selection bias*;<sup>12</sup> ii) each vocabulary is found to capture the multifaceted characteristics of each climate risk type, rather than single aspects, *complete climate risks information*;<sup>13</sup> iii) to the best of our knowledge, this is the first study distinguishing between physical and transition risk while acknowledging that these share common concepts, *risk interconnections are context-scaled*; iv) the ability to rank terms by relevance allows for a deeper understanding of each risk type and enables to examine which risk aspects are more or less important in the overall risk description, *relevance-ranked vocabularies*;<sup>14</sup> v) the resulting climate risk vocabularies represent global physical risk and transition risk vocabularies, because built

and because they can be considered as outliers due to their great length and very marginal occurrence.

<sup>&</sup>lt;sup>11</sup>The final collection of documents is then composed by T documents, as a total of T-1 daily news documents and 1 physical (transition) risk document, which enables us to calculate the *idf* scores. At this stage, to lighten the computational load and avoid the so-called machine learning overfitting issue, we consider a subsample of the Reuters News (2015-2019).

<sup>&</sup>lt;sup>12</sup>Some studies initially use keywords defined by authors to construct climate indicators (e.g. Faccini et al. 2023) or news provider *tags* (e.g. Ardia et al. 2023).

<sup>&</sup>lt;sup>13</sup>Previous research mainly identifies only sub-categories of climate risks.

<sup>&</sup>lt;sup>14</sup>Other studies rely mainly rely on a term-frequency ranking.

on scientific information, and can be applied to any type of text to assess the its level of physical risk or transition risk discussion, *global climate risks vocabularies*.

Figure 1 shows the most relevant terms of the physical risk vocabulary (a) and the transition risk vocabulary (b) as word clouds, where each term size is proportional to its *tf-idf* score. The physical risk vocabulary includes multiple dimensions of physical risk such as both extreme and chronic hazards directly caused by climate change, excluding natural disasters attributable to other sources. Accordingly, the transition risk vocabulary includes various aspects of this climate risk such as technological advances and environmental policies. Terms such as "ecosystems", "sea level", and "precipitation" are representative of the physical risk topic, while terms such as "hydrofluorocarbon" (HFC), "bioenergy", and "greenhouse gas" (GHG) are representative of the transition risk topic. Additionally, the figure shows evidence that the constructed vocabularies are likely to capture interconnections between the two complex risks, and to contextualise common terms. For instance, the term "GHG" appears in both vocabularies, but to a different extent. It plays a primary role in explaining transition risk and a minor one for physical risk. The term "adaptation", on the other hand, is a common concept for physical and transition risk and therefore appears in both vocabularies. However, its semantics differ depending on whether it is being considered within the context of physical or transition risk and thus depends on the other terms in the vocabulary.

Nevertheless, for the exercises to be carried out in this paper, it has to be assured that vocabularies are sufficiently different in terms of the content they describe. To confirm that this is the case, we apply a test proposed by Dang et al. (2015).<sup>15</sup> Results show that the transition risk vocabulary is able to explain less than 5% of the physical risk vocabulary, which in turn carries about 95% of individual information, and vice-versa.

#### - Insert Figure 1 about here -

<sup>&</sup>lt;sup>15</sup>We evaluate the actual degree of commonality between the two vocabularies as the R squared from regressing the physical risk vocabulary on the transition risk one, and vice-versa. Despite there not being a clear threshold level (Rognone et al. 2020), the resulting R square of less than 5 percent is considered sufficiently small to support a reliable separation of the two risks.

#### 2.2. Physical and Transition Risk Indices

In order to calculate our two risk indices we first compute two "concern" series. These news media concern series for physical/transition risk, on any given day *t*, are defined as the cosine-similarity between the *tf-idf* vector of the news document and the physical/transition risk document. The cosine-similarity is used in text-analysis to evaluate the similarity between pairs of texts. It expresses the angular distance between two vectors representative of pairs of text, such that the closer they point, the smaller their angular distance, the higher the cosine, and the higher the text similarity. In other words, we consider our physical (transition) risk dictionary as a vector, the direction of which depends on the intensity of each element given by the *tf-idf* of vocabulary terms. This means that daily news that point in the same direction as the physical (transition) risk vector are assessed to discuss the physical (transition) risk topic and the concern score, or cosine-similarity score, roughly represents the portion of daily news corpus dedicated to the topic of physical (transition) risk.

In order to gauge the unexpected change in physical/transition risk, we then construct the Physical Risk Index (PRI) and the Transition Risk Index (TRI) as residuals from autoregressive processes of order 1 (AR1) similarly to Engle et al. (2020), as follows

$$Concern_{t,PR} = c_{PR} + \phi_{PR}Concern_{t-1,PR} + PRI_{t,PR}$$
(1a)

$$Concern_{t,TR} = c_{TR} + \phi_{TR}Concern_{t-1,TR} + TRI_{t,TR}$$
(1b)

Table 1 reports a summary of the ten highest physical and transition risk shock days, showing the dates, PRI and TRI values, and news topics and titles of the most relevant articles. High shock days can cover a multiplicity of physical and transition risk topics. On one hand, the PRI is found to capture not only acute physical risks such as floods, or extreme weather events, but also a plurality of chronic risks such as permafrost thawing, droughts, and sea level rise, as well as governments and other institutions calls for climate adaptation actions and other adverse impacts of physical risk on, e.g., the ecosystem such as biodiversity loss. This sets our PRI apart from many other physical risk databases, which mainly identify extreme weather events only. On the other hand, news on regulations and measures to curb GHG emissions generate large spikes in the TRI, e.g. news regarding the EU carbon reform deal or the Montreal Protocol, as well as news concerning the costs associated with the transition or the advances of technological innovation and renewable energies to reach, e.g., net-zero emissions targets.

The PRI peak is registered on 19/09/2018, mainly due to an unexpected discussion about an unprecedented loss of arctic sea ice underling a critical level of permafrost thawing risk, but also due to discussions about sea level rise and chronic changes in the salinity of oceans. The largest shock for TRI concurs instead with news published on 24/08/2011, which mainly covered the worryingly high levels of EU GHG emissions which would need to be reduced.

#### - Insert Table 1 about here -

Figure 2 shows the scatter plots of daily physical and transition media concerns along with major PRI and TRI topics (panels a and b), and their monthly average (panels c and d). Table 2 summarises the AR1 estimates from Equations (1a) and (1b). Both physical risk and transition risk concern time series depict positive drifts ( $c_{TR} = 8.462\%$  and  $c_{PR} = 7.862\%$ ), showing that the news coverage toward these climate risks tends to increase over time. The media concern for transition risk seems to be more persistent than that for physical risk with  $\phi_{PR} = 0.326$  and  $\phi_{TR} = 0.413$ . On the other hand, while the risk indices PRI and TRI are positively correlated with a coefficient of 0.37, according to the Dang et al. (2015) commonality test the 86.22% of the PRI and TRI information represents individual information, ensuring an adequate separation of physical and transition risk. In fact, only the 13.78% of PRI can actually be explained by the TRI series, and vice-versa. Because the two risks are naturally interconnected, as for instance one cannot explain the full transition risk without at least mentioning the physical risk, a certain degree of overlap is expected. Additionally, the same news story can sometimes discuss more broadly climate change covering both climate risk types.

- Insert Figure 2 and Table 2 about here -

Positive, zero, or negative values of our climate risk indicators signify an above-expected,

expected, or below-expected level of discussion around climate risk issues (either physical or transition), thus representing a positive, null, or negative innovation, respectively. Essentially, because the TRI and PRI represent innovations/shocks from the AR1 process of the so-called concern series, they can be interpreted in this manner. Theoretical underpinnings, including insights from IPCC reports and existing climate finance literature, strongly support the notion that climate risks represent a source of financial risks. Thus, asset prices should reflect climate risk information, although potential mispricings may arise due to, e.g., the effectiveness of measuring climate risk and exposure, or varying degrees of scepticism regarding climate change issues. Given the ability of our novel PRI and TRI to capture various climate risk aspects, we consider these as new pricing factors to conduct our empirical analysis.

# 3. Transition and Physical Risk Pricing in European Equity Markets

In the following section, we present the methodological approach to examine the existence of physical and transition climate risk premia in equity markets exploiting the physical and transition risk indicators. Afterwards, we discuss how we investigate which information type may be used by investors to proxy firms' exposure to either physical or transition risk.

#### 3.1. Transition and Physical Climate Risk Premia

To assess the presence of climate risk premia for physical and transition risk in the crosssection of European stock returns we adopt a standard portfolio sorting approach. We perform time-series regressions of equity returns on our climate risk indices, controlling for standard risk factors known to drive returns. The resulting loading on the climate risk factors, i.e. the transition and physical risk betas, are our firm-level indicators of climate risk exposure. We use these to sort companies from low to high climate exposure and create portfolios. The negative/low TRI (PRI) beta portfolio includes firms that perform badly when there are increases in transition (physical) risk. Conversely, the positive/high beta portfolio includes firms that perform well when transition (physical) risk rises. Since investors can be expected to want to hedge against climate risks, they should be willing to accept lower expected returns for equities that offer good performance in case of bad transition (physical) risk economy states, i.e. for high beta stocks. In turn, low beta stocks should instead trade at a discount and offer higher expected returns to compensate for the higher risk investors bear to hold such climate riskier stocks in their portfolios. A low-minus-high transition (physical) climate beta portfolio should therefore earn positive abnormal excess returns in the case a climate risk premium exists.

Specifically, at the end of every month, we recursively estimate the sensitivity of each stock i to both transition and physical climate risks (climate betas) adopting a rolling window of daily observation over the past three months and controlling for the standard Fama and French (2015) five factors (FF5) as follows

$$r_{i,t}^{exc} = c_i + \beta_i^{TRI} \operatorname{TRI}_t + \gamma_i^{TRI} X_t + \epsilon_t$$
(2a)

for transition risk, and

$$r_{i,t}^{exc} = c_i + \beta_i^{PRI} \mathbf{PRI}_t + \gamma_i^{\prime PRI} X_t + \epsilon_t$$
(2b)

for physical risk, where  $r_i^{exc}$  denotes the daily excess return on security *i*,  $c_i$  is the constant term,  $PRI_t$  and  $TRI_t$  are our climate risks measures, and the vector  $X_t$  controls for the FF5 factors known to drive the cross-section of stock returns, namely the market, the size, the book-to-market, the profitability, and the investment factors.<sup>16</sup> It could be anticipated that companies whose business is, e.g., adverse to new climate regulations or reliant on climate stability may be characterised by a negative transition or physical climate beta, respectively.

We sort stocks according to their estimated betas at the end of each month and group them into 5, 10, and 25 portfolios for which we compute the post-ranking equal-weighted monthly returns.

<sup>&</sup>lt;sup>16</sup>The market factor represents market returns above the risk-free rate, the size factor measures additional returns from small-cap stocks, the book-to-market or value factor tracks excess returns for lower-valuation companies, the profitability factor identifies extra returns from highly profitable companies, and the investment factor reflects surplus returns from conservative investment strategies versus more aggressive ones. The Fama and French (2015) five factors are constructed considering the EuroStoxx 600 Index constituents over the period Jan 2005-Oct 2021 for which we calculate the 6 value-weight portfolios formed on size (market capitalization) and book-to-market, the 6 value-weight portfolios formed on size and operating profitability, and the 6 value-weight portfolios formed on size and investment (change in total assets). Data are collected from Eikon. More details on the methodology can be found in the Kenneth R. French' data library.

To examine the TRI-return relation (PRI-return relation), we also form a low-minus-high (LMH) portfolio that takes a long position in the negative-beta TRI (PRI) portfolio and a short position in the positive-beta TRI (PRI) portfolio, and we calculate the returns on this portfolio. We evaluate the transition (physical) risk premium, estimating each LMH climate portfolio alpha while considering the Fama and French (2015) five factors asset pricing model specification as follows

$$r_{\beta TRI,t}^{exc} = \alpha_{TRI} + \gamma_i^{\prime TRI} X_t + \epsilon_t \tag{3a}$$

for transition risk, and

$$r_{\beta PRI,t}^{exc} = \alpha_{PRI} + \gamma_i^{\prime PRI} X_t + \epsilon_t \tag{3b}$$

for physical risk.

Aligned with recent studies which find larger importance of climate risks for financial markets over the past few years (Goldsmith-Pinkham et al. 2023; Krueger et al. 2020; Painter 2020; Bolton and Kacperczyk 2023, 2021), especially from the time of the Paris Agreement, we allow our study to capture potential changes in the relation between European equity markets and climate risks by looking both at the full sample period and at the period before and after 2015.

#### 3.2. The Use of Climate Exposure Metrics by Investors

In this paper we refer to "metric" as any type of information investors may use to assess firms' exposure to climate risks. Given the lack of a unique metric, investors can be expected to make use of different types of information to identify climate-risky stocks.<sup>17</sup> We reckon that while most of the common metrics are used to capture exposure to transition, rather than physical risk, the distinction is not always clear and their potential to capture physical exposure has been largely unexplored. For this reason, we decided to test a wide range of exposure metrics in light of their potential use by investors.

We add the Physical Risk Index (PRI) and the Transition Risk Index (TRI) to a Fama and French (2015) five factors (FF5) asset pricing model. We consider the E score, ESG score, GHG emissions level, and GHG emissions intensity as *firm-level* exposure metrics to sort firms and

<sup>&</sup>lt;sup>17</sup>A discussion of the most common metrics is featured in Appendix A2.

create green and brown portfolios, as follows:

- *E score* and *ESG score* metrics. Firms whose E score is above (below) the 75th (25th) percentile are defined as green (brown). The green (brown) portfolio is then created as an equally weighted portfolio composed of green (brown) firms. The same approach is applied to the ESG score metric. Portfolios are rebalanced annually;
- *GHG emissions level* and *GHG emissions intensity* metrics. The GHG emissions level  $(GHG_E)$  is calculated as the sum of Scope 1 and 2, while the GHG emissions intensity  $(GHG_{EI})$  is calculated as the GHG emissions level scaled by firms' net revenue. As before, firms whose emissions level is below (above) the 25th (75th) percentile are defined as green (brown) firms. Portfolios are again rebalanced annually.

At the same time, we conduct a *sectoral-analysis* by aggregating the excess returns of firms belonging to the same sector (NACE Rev. 2) to test the aggregate response of different sectors to shocks to physical and transition risks.

We therefore include our TRI and PRI into a model to gauge equity excess returns

$$r_{p_i,t}^{exc} = c_{p_i} + \beta_{p_i}^{TRI} \mathbf{TRI}_t + \gamma_{p_i}^{\prime TRI} X_t + \epsilon_t$$
(4a)

to price transition risk, and

$$r_{p_i,t}^{exc} = c_{p_i} + \beta_{p_i}^{PRI} \mathbf{PRI}_t + \gamma_{p_i}^{PRI} X_t + \epsilon_t$$
(4b)

to price physical risk. For the firm-level analysis,  $r_{p_i}^{exc}$  denotes the excess return at time t for green or brown portfolios where  $p = \{\text{green portfolio}, \text{ brown portfolio}\}\)$  and  $i = \{\text{GHG}_E, \text{GHG}_{EI}, \text{E}, \text{ESG}\}$ . For the sectoral analysis,  $r_{p_i}^{exc}$  instead denotes the excess return at time t for the portfolio p that aggregates firms of sector i.  $c_{p_i}$  is the constant term and the vector  $X_t$  controls for the FF5 factors. The coefficients  $\beta^{PRI}$  and  $\beta^{TRI}$  measure the relationship between an unexpected change in physical and transition risk, and the excess returns of portfolios constructed according to different exposure metrics. The results from this exercise could inform us about the metric used by investors to proxy firms' exposure to physical or transition risk.

## 4. Data

The augmented FF5 model uses the 1-month Overnight Index Swap (OIS) rate as the risk-free rate, and returns of the EuroStoxx 600 Index as a proxy for the market return. All data is used at a daily frequency. We collect closing price time series for the historical constituents of the Eurostoxx 600 Index from Datastream over the period Jan 2005-Oct 2021, resulting in a total of 1,198 companies.

Data on firms' GHG emissions level, GHG emissions intensity, E score, and ESG score are sourced from Refinitiv. The level of GHG emissions indicates the metric tonnes (in thousands) of carbon dioxide equivalent a company produces.<sup>18</sup> We compute the GHG emissions intensity as GHG emissions scaled by the firm's net revenue. The E score rather reflects the environmental performance of a company in terms of its commitment and effectiveness to tackle issues related to the use of resources, emissions, and innovation, while the ESG score is also informative about a firm's performance concerning social and governance issues. E and ESG scores are industry-based relative performances, relative to sector peers. We define sectors using the Statistical Classification of Economic Activities in the European Community (NACE Rev. 2).<sup>19</sup> Table 3 shows the number of firms in our sample with available GHG emissions, GHG emissions intensity, E scores, and ESG scores data, highlighting a general increase in data coverage over time. The table also reports the threshold values (25<sup>th</sup> and 75<sup>th</sup> percentiles) used to construct brown and green portfolios for each metric.

#### - Insert Table 3 about here -

<sup>&</sup>lt;sup>18</sup>The GHG Protocol Accounting and Reporting Standard classifies a company's GHG emissions into three scopes (Schmitz et al. 2004). We measure GHG emissions as the sum of scope 1 (direct emissions from company-owned and controlled resources) and 2 (indirect emissions from purchased electricity by the owned or controlled equipment or operations of the firm) because including scope 3 (other supply chain emissions) reduces the data coverage. We consider only data reported by the company.

<sup>&</sup>lt;sup>19</sup>Eurostat (2008). Dafermos et al. (2020) for example identifies high-carbon intensive activities taking NACE 1-digit sectors that mostly contribute to EU emissions.

#### 4.1. Descriptive Statistics

In order to give a better overview of the composition and characteristics of the EuroStoxx 600 Index at the sectoral level, table 4 reports the number of firms in our sample (No.), the average of the exposure metrics (E, ESG, log-GHG<sub>EI</sub>, log-GHG<sub>E</sub>) and the yearly average contribution of each sector to the GHG emissions of the EuroStoxx 600 Index (GHG<sub>E</sub> contribution Index) and EU (GHG<sub>E</sub> contribution EU).<sup>20</sup> The table is sorted by GHG emissions in descending order, with lighter (darker) colours being associated with greener (browner) sectors.

As expected, D-Electricity, gas, steam and air conditioning supply, C-Manufacturing, and H-Transportation and storage are among the most GHG emitting sectors, contributing around 70% of total EU emissions and 55% of total EuroStoxx 600 Index emissions, respectively. In comparison, the A-Agriculture, forestry and fishing is a high emissions contributor at the European level (16%), but not in our sample (0%), possibly due to a low representation of companies from this sector in the EuroStoxx 600 Index. B-Mining and quarrying and M-Professional, scientific and technical activities<sup>21</sup> are instead minor GHG contributors at the European level but major in our sample. Additionally, sectors with good average E (and ESG) ratings also have, on average, high GHG emissions levels (and intensity) suggesting that companies with high GHG emissions can receive positive environmental and ESG scores, and vice versa (Boffo et al. 2020). Positive E, or ESG, ratings are therefore not necessarily associated with low carbon emissions, aligned with the assumption that E and ESG can capture aspects of climate risk further to GHG production (Faccini et al. 2023).

#### – Insert Table 4 about here –

Figure 3 presents the distribution of the four metrics used. E and ESG scores appear quite homogeneous across sectors, while GHG emissions largely differ within and across sectors. This is in line with the fact that Refinitiv ESG scoring methodology is aimed at reducing portfolio

<sup>&</sup>lt;sup>20</sup>EU27, Data source: Eurostat.

<sup>&</sup>lt;sup>21</sup>The broad characterization of this sector makes its interpretation challenging. In our sample 70 percent are activities carried on by head offices.

concentration by sectors and thus recalibrates upwards the rating of polluting companies if they are in high polluting sectors, i.e. a company is largely evaluated relative to its sector peers. The distribution of GHG emissions by sectors also shows that the NACE classification may not take into account emissions-related intra-sectoral differences.

Finally, Table 5 shows the sectoral composition of brown and green portfolios according to the different exposure metrics used. We can observe that the composition of the brown (green) E portfolio is very similar to that of the brown (green) ESG portfolio, while the composition of the brown (green) GHG emissions portfolio shares similarities with the brown (green) GHG emissions intensity portfolio. However, the portfolios constructed according to E or ESG criteria differ significantly from the ones based on GHG emissions. This is in line with the observation that high GHG emitting companies can receive high ESG and E scores.

- Insert Figure 3 and Table 5 about here -

## 5. Results

Results provide evidence of economically significant transition risk and physical risk premia, as sizable as 7.05% and 6.14% from 2015 onwards respectively.<sup>22</sup> Our findings also suggest that, when stocks are sorted according to common metrics to evaluate the "greenness" and "brownness" of firms (such as GHG emissions, ESG score), rises in transition risk typically increase the excess returns of green stocks, while rises in physical risk typically decrease the excess returns of brown stocks. The sectoral analysis reveals that sectoral information may not or no longer be granular enough to capture transition risk exposure. Rather, sectoral characteristics may be used by investors to identify exposures to physical risk.

<sup>&</sup>lt;sup>22</sup>While increasing over time, the statistical significance of such risk premia remain below the usual acceptance levels for the time period studied.

#### 5.1. Transition and Physical Climate Risk Premia

Results indicate the emergence of both a physical and a transition climate risk premium since 2015, providing supporting empirical evidence that investors are already demanding extra compensation for exposure to climate-related risks and that such compensation increases in the level of exposure. Table 6 presents the annualized average excess stock returns in percentage (E[R]-Rf, in excess of the risk free-rate), standard deviations, and Sharpe ratios of the quintiles, deciles and 25 percentiles L, H, and LMH portfolios for transition (Panel a) and physical (Panel b) risk, over the three periods studied (full sample, before and after 2015). The table shows that the 25 percentiles LMH transition (physical) generates an average annualized return of about -3.08% (-3.75%) in the period before 2015 and 9.61% (6.71%) after 2015. The finding that the returns on the LMH portfolios are economically large and increasing over time suggests that firm-level sensitivity to both TRI and PRI may have predictive ability for stock returns. In the case where we consider decile and quintile portfolios, the evidence for increased performance also holds, but to a lower extent. Figure 4 shows the cumulative performances of the LMH transition and physical quintiles (lightgrey), deciles (dark-grey), and 25 percentiles (black) portfolios highlighting a very rapid increase in returns after 2015 for the 25 percentiles LMH portfolio. As in table 6, the figure shows stronger effects for more extreme portfolio specifications, i.e. from 5 to 25 percentiles.

- Insert Table 6 and Figure 4 about here -

Table 7 presents the estimated physical and transition climate risk premia, i.e. alphas, from the FF5 factors model for the quintile, decile, and 25 percentile L, H, and LMH portfolios sorted on PRI (Panel a) and TRI (Panel b) over the full sample period, before and after 2015. These estimates document the emergence of physical and transition risk premia since 2015, with the economic significance being stronger for more extreme portfolios. This result implies that the compensation investors require for holding physical and transition risky stocks depends on the degree of climate risks they expose themselves to, demanding higher compensation for higher exposure, e.g. the 25 percentiles LMH portfolio generates a higher alpha than that from the quintiles LMH portfolio.

The 25 percentiles long-short PRI and TRI portfolios are found to generate an average abnormal return of about -4.09% and -3.01% per year before 2015, and of about 6.14% and 7.05% after 2015, respectively. The finding of an elevated transition risk premium is consistent with Bolton and Kacperczyk (2021) who, however, focus on GHG emissions in their analysis. Therefore, our findings expand on Bolton and Kacperczyk (2021) since our measure of transition risk goes beyond carbon emissions and encompasses various sources of transition risk. Consequently, we contribute by demonstrating that the comprehensive transition risk premium is increasing over time<sup>23</sup> Additionally, we contribute to the literature as we observe a similar trend in the physical climate risk premium. This result contradicts findings from, for instance, Faccini et al. (2023) who did not find any pricing associated with the direct manifestations of physical climate change. Faccini et al. (2023) find that investors only hedge against imminent transition risks resulting from government actions to combat climate change. In contrast, our results reveal that, at least in Europe, investors are seeking compensation for both exposure to transition risks and physical risks, in line with the theoretical expectations that climate risks should be incorporated into asset prices.

While our findings do share common ground with Bolton and Kacperczyk (2021) regarding the presence of an elevated transition risk premium, it's important to note a distinction. Bolton and Kacperczyk (2021) identify Asia as "entirely responsible for the rise in the global carbon premium around the Paris agreement" while measuring transition risk as one single factor, i.e. GHG firms emissions. However, our results, which instead consider a more complete measure of transition climate risk, show that a transition risk premium has sharply risen in Europe too, implying that any post-Paris Agreement global transition premium can probably not solely be attributable to Asia. This further demonstrates that empirical results might be sensitive to the use of climate measures, stressing that climate risk measurement needs precision to unveil economically meaningful implications of climate risks.

 $<sup>^{23}</sup>$ Compared to Hsu et al. (2023) who find that long-short portfolio constructed from firms with high versus low toxic emission intensity generates an average excess return of around 4.42% per year over the period 1992-2018, our findings depict a higher transition risk premium. It is worth noting that the transition measures, sample periods, and regional focus are different.

#### – Insert Table 7 about here –

#### 5.2. Portfolios Sensitivity to Climate Risks

#### 5.2.1. Firm-level analysis

Table 8 reports the results for the estimated factor sensitivities of green and brown portfolios constructed according to E scores, ESG scores,  $GHG_{EI}$ , and  $GHG_E$  from the daily augmented FF5 model as presented in Equations (4a) and (4b) over three periods (full sample, before 2015, and after 2015) reported together with t-statistics and considering Newey and West (1987) robust standard errors.<sup>24</sup>

The table provides insights that both E/ESG and GHG emissions intensity data appear to be useful gauges for investors to identify firms positively exposed to transition risk. We find evidence that the excess return of the green ESG portfolio significantly increases as transition risk rises both before and after 2015, of about 0.308 and 0.345% respectively, and that such increase is robust also when considering the full sample period. This suggests that unexpected changes to transition risk have been incorporated into asset prices as far as 2005, such that firms with good ESG scores have performed well during days where TRI increased, and that ESG-oriented investors have earned positive realized returns during high transition risk days. Additionally, green E and GHG<sub>E1</sub> portfolios significantly increase in returns of about 0.330 and 0.521%, respectively, when the market is surprised by transition risk news post-2015. Finally, we do not find evidence for an effect of physical risk on green portfolios.

Concerning brown portfolios sensitivity to climate risks, there is no evidence of a significant decrease in returns of brown portfolios in relation to rises to transition risk. However, there is evidence of a statistically significant negative relationship between the excess returns of brown  $GHG_{EI}$  and  $GHG_E$  portfolios and rises in physical risk since 2015, so that firms with a high GHG emission net-revenue ratio drop in returns of about 0.921% a day due to increases in PRI, and firms with high GHG emissions levels decline in returns of about 0.765%.

<sup>&</sup>lt;sup>24</sup>We use Newey-West standard errors throughout.

By separating physical and transition risk, we are able to refine previous findings from the green finance literature, specifically those of Ardia et al. (2023) and Pástor et al. (2022), by identifying transition risk, rather than physical risk, as the main source of climate risk for green stocks. Additionally, while Ardia et al. (2023) document that unexpected increases in climate change concerns decrease the returns of brown firms, we provide empirical evidence that it is the physical climate risk component, rather than the transition risk one, which drives such a result. Without a distinction between climate change risks, in fact, the previous literature could only determine the relationship between aggregate climate change risk and green/brown stocks returns ignoring if results are driven by the physical or the transition risk component, information that would be instead useful for, e.g., tailored hedging strategies.

#### 5.2.2. Sectoral analysis

Turning to the sectoral classification as a gauge for firms' climate risk exposures, tables 9 and 10 show regression results for the three sample periods (full sample, before 2015, and after 2015), using the NACE sectoral classification to group excess returns of European companies examining transition and physical risk respectively. This exercise and the relative findings further contribute to the work by Bolton and Kacperczyk (2021), who question whether investors consider the industry where firms operate in, rather than firm-level information, as material information on firms' exposure to climate risks.

Table 9 shows that coefficients for TRI are largely insignificant post-2015. This suggests that sectoral information may not or no longer be granular enough to capture transition risk exposure, possibly due to the increased uncertainty about the future course of climate-related policies (Batten et al. 2016) and their impact on sectors and industries. Rather, investors may rely on firm-level information such as ESG or E ratings, as we find in the firm-level analysis, to proxy firms' sensitivity to transition risk. Table 10 shows that, after 2015, PRI coefficients for sectors which are expected to be exposed to physical risk are found to be negative and significant. In particular, we document that the excess returns of the B-Mining and quarrying sector, the H-Transportation and storage sector, and the J-Information and communication sector significantly decrease by about

-2.42, -1.06, and -0.70%, respectively, in relation to rises in physical risk after 2015. These sectors are exposed to physical risk through their infrastructure assets or natural systems such that these activities suffer losses from, e.g. interruptions of operational activities due to physical hazards. Interestingly, the K-Financial and insurance activities sector is found to perform well during days in which news unexpectedly discuss issues related to physical climate risk over the full sample period (0.326%), and before 2015 (0.475%), but no significant effect is documented from 2015 onwards. We also find a negative but not statistically significant relationship between physical risk and the excess returns of L-Real estate activities after 2015. This result appears aligned with the Murfin and Spiegel (2020) belief of optimism about the physical chronic hazard of rises in sea levels or the possibilities of mitigation and bailouts, and contrasts with Bernstein et al. (2019) and Baldauf et al. (2020). Overall, findings suggest that sectoral characteristics may be used by investors in evaluating firms and activities' exposure to physical risk.

#### 5.3. Physical and transition climate risks: A country-level outlook

Climate-related risks are expected to affect different countries in a heterogeneous manner depending on a multiplicity of factors, such as the level of a country's investment in climate resilient infrastructure for, e.g., floods or other climate events and/or the state of national and sub-national climate and environmental policies, making a comprehensive assessment of individual countries' exposure to physical and transition risk difficult to estimate (Bolton and Kacperczyk 2023). The economics literature on climate change finance further underscores the importance of the geographical distribution of climate policies (Nordhaus and Yang 1996) and physical risks (Cruz and Rossi-Hansberg 2023).

In this section, we aim to provide insights on European individual countries' sensitivity to climate-related risks over time by employing our new climate risk factors as an explanatory variable for countries' stock market indices as follows

Stock Market Index Country<sub>*i*,*t*</sub> = 
$$\alpha + \beta_i CCR_{j,t} + \varepsilon_t$$
 (5)

where Stock Market Index Country<sub>*i*,*t*</sub> is the log-return of the stock market index for the European country *i* at time *t*,  $CCRI_{j,t}$  is our climate change risk index with  $j = \{$ physical risk, transition

risk} at time t. Thus,  $CCRI_{\text{physical risk},t} = PRI$  and  $CCRI_{\text{transition risk},t} = TRI$ .  $\alpha$  and  $\beta$  are model parameters and  $\varepsilon$  is the error term.

Figure 5 displays bar plots representing the Ordinary Least Square (OLS) estimates of  $\beta$  for transition (a) and physical (b) risk factors before 2015 (light grey) and after 2015 (dark grey) for each country within the sample. The significance levels are indicated with one, two, or three asterisks, corresponding to p-values below 0.01, 0.05, and 0.1, respectively. The key takeaways from figure 5 are threefold. First, the sensitivity to climate risks varies country by country both in terms of magnitude and statistical significance. Second, physical and transition risks are confirmed to exert different effects on stock markets. Third, there is heterogeneity in results depending on whether we consider the period before or after 2015. Overall, our results suggest that a country-level connection between the stock market and climate risks exists, while this result is heterogeneous by country further underscoring the importance of the consideration of sub-national conditions in both investment decision-making and when designing climate policies.

Figure 6 illustrates the results in an alternative form, i.e. as coloured heated geographical maps with colder/bluer colours indicating positive country sensitivity to climate risk and warmer/redder colours indicating negative sensitivity. The figure helps to gain an understanding of the evolving exposure of countries to transition and physical risks. Panels a and c show Europe's sensitivity to transition and physical risk on a per-country basis before 2015, while panels b and d provide the findings for the period after 2015. Notably, there is a discernible shift in the overall sensitivity to climate risks between the pre-2015 and post-2015 periods, with the latter being characterized by a more negative overall association with climate risks. In fact, despite some heterogeneity, panels b and d are predominantly shaded in reddish tones compared to panels a and c, indicating that countries are experiencing more pronounced impacts from both transition and physical risks, possibly due to the implementation of stricter climate regulations and the heightened severity and frequency of physical climate events. As an example, Greece emerges as the country most negatively associated with transition climate risks after 2015, indicated by a statistically significant beta of approximately -4. This finding underscores how Greece's stock market is particularly suscep-

tible to risks linked to the shift toward a more environmentally sustainable economy. Meanwhile, Norway stands out as Europe's most vulnerable nation to physical climate risks post-2015, characterized by a highly statistically significant beta of about -3. This observation highlights how the country suffers from tangible manifestations of climate-related hazards, which also have adverse repercussions on its entire stock market.

These findings highlight the importance of incorporating country-specific attributes into the design and development of climate and environmental policies. As climate risks affect countries in distinct ways, neglecting these disparities could have implications for financial stability. Furthermore, we observe that the impact of physical risks on countries' stock markets can yield diverse outcomes. On one side, countries are inherently exposed to distinct physical risks; for instance, one nation might be more susceptible to flood hazards than another to wildfires (Bolton and Kacperczyk 2023; Cruz and Rossi-Hansberg 2023). These inherent distinctions may lead to varied responses to physical risks. On the other side, several other factors can also shape a country's sensitivity to physical risks, such as its capacity to manage and adapt to climate-related events. Thus, our findings suggest, for instance, that improved government-led adaptation measures might take into account country-level characteristics. Finally, from an investor's perspective, our results imply that incorporating country-specific information into investment decision-making can be advantageous to manage both transition and physical risks.

- Insert Figures 5 and 6 about here -

#### 5.4. Policy implications

In addition to advancing our understanding of climate risks in financial markets, our findings also carry important implications for policymakers.

First, the identification of transition and physical risk premia in European equity markets highlights the need for integrating climate risks into the core of financial risk management and regulatory supervision in so far as possible. It becomes crucial for financial institutions to factor in climate risk premia when making portfolio allocation decisions to account for potential shifts in asset valuations due to climate-related events or transition-related policy changes. Regulatory bodies, including the European Central Bank, play a guiding role and, when necessary, mandate the consideration of these climate risks in stress-testing exercises, thereby safeguarding the stability of the financial system.

Second, our findings underscore the importance of firm-level data in gauging a company's exposure to climate risks, particularly those linked to the transition to a greener economy. This suggests that there is value in advocating for enhanced transparency and consistency in firms' disclosure of environmental metrics, such as GHG emissions and ESG performances. Improved disclosures enable investors to make more informed decisions, and regulatory bodies have a pivotal role to play in standardizing and ensuring the accuracy of these disclosures, thereby promoting a more efficient allocation of capital in the markets and reducing the risk of climate-related financial bubbles or shocks.

Third, our results indicate that certain sectors are particularly sensitive to (physical) climate risks from an investor perspective. Recognizing and supporting sectors vulnerable to (physical) risks through infrastructure resilience programs or R&D in sustainable alternatives can contribute to fostering more sustainable economic growth. Conversely, sectors exhibiting resilience or positive performance despite increasing physical risks could be further studied and integrated into broader strategies to combat the economic implications of climate change.

### 6. Conclusions and Directions for Future Research

The role of climate-related risks in financial markets has attracted large attention from financial market participants, governments, and academics. Investors want to know the asset prices sensitivity to climate change risks and regulators are concerned about the consequences of an incorrect pricing of climate-related risks. Against this backdrop, this paper contributes to our understanding of the financial implications of climate-related risks, exploring whether, and to which extent, physical and transition risks are incorporated into asset prices.

Using text-analysis, we first build two distinct vocabularies for physical and transition risk

which derive from authoritative and scientific sources. The constructed vocabularies benefit from several advantages, improving on the existent literature. Then, using the cosine-similarity approach, we compare them to a corpus of news to obtain a novel comprehensive Physical Risk Index (PRI) and Transition Risk Index (TRI). PRI and TRI are found to spike during days of high discussion about either the topic of physical or transition risk, capturing multiple aspects of these risks. PRI and TRI are then employed to examine the existence of physical and transition climate risk premia in European equity markets.

Our results have important economic implications and show the emergence of both economically significant transition and physical risk premia in European markets as sizable as 7.05% and 6.14% on average per year after 2015, respectively. Using our comprehensive climate risk measures, we show that a transition risk premium has arisen in Europe, challenging Bolton and Kacperczyk (2023) who, using only GHG emissions to measure transition risk, find that Asia is entirely responsible for a global transition premium around the Paris Agreement. We provide empirical evidence supporting that (European) investors are demanding extra compensation for their exposure to both transition and physical risks, and that such compensation increases in the level of exposure. We then test the sensitivity of portfolios constructed according to common firm-specific climate risk exposure metrics (e.g. ESG scores and GHG emissions) to our climate risks indicators. Then, comparing the results to those from a sectoral analysis, we are able to investigate whether investors may simply pigeonhole firms into the industry they operate in, rather than use firm-level information, to detect climate-risky stocks. On one hand, findings show that firm-level information appears to be used by investors as a gauge for firms' exposure to climate risks, in particular for transition risk and since 2015. Additionally, by distinguishing physical from transition risk we can better describe the relationship between green/brown stock returns and climate risks, with respect to the existent literature. In this regard, we refine the Ardia et al. (2023) and Pástor et al. (2022) findings showing that rises in transition (physical) risk typically increase (decrease) the return of green (brown) stocks. On the other hand, our results suggest that the sectoral classification may be employed to broadly proxy firms' exposures to physical risk. Finally, from a per-country basis analysis, we observe that European stock markets and climate risk connection have intensified post-2015, yet with some heterogeneity emphasizing the need for national/sub-national considerations in investment and climate policies.

The findings presented in this paper, with the most important contribution being the Transition Risk Index (TRI) and Physical Risk Index (PRI), inform investors, policymakers, and financial institutions about the extent to which financial markets price climate risks and contribute to the on-going debate on how to incorporate climate risks into risk management considerations. Additionally, our climate risk indices can be used to study the role of climate risks for other asset classes, and more broadly find applications to risk and portfolio management issues.

# References

- Addoum, J. M., Ng, D. T., and Ortiz-Bobea, A. (2020). Temperature Shocks and Earnings News. Review of Financial Studies, 33, 1331–1366.
- Adomako, S., Simms, C., Vazquez-Brust, D., and Nguyen, H. T. (2023). Stakeholder Green Pressure and New Product Performance in Emerging Countries: A Cross-country Study. British Journal of Management, 34, 299–320. URL: https://onlinelibrary.wiley.com/doi/10.1111/1467-8551.12595.doi:10.1111/1467-8551.12595.
- Alessi, L., Ossola, E., and Panzica, R. (2021). What greenium matters in the stock market? The role of greenhouse gas emissions and environmental disclosures. Journal of Financial Stability, 54, 100869. URL: https://linkinghub.elsevier.com/retrieve/pii/ S1572308921000280. doi:10.1016/j.jfs.2021.100869.
- Ali, S., Badshah, I., Demirer, R., Hegde, P., and Rognone, L. (2023). Climate Uncertainty and Investor Learning in Sustainable Funds. SSRN Electronic Journal. doi:10.2139/ssrn. 4432798.
- Ardia, D., Bluteau, K., Boudt, K., and Inghelbrecht, K. (2023). Climate change concerns and the performance of green vs. brown stocks. Management Science, 69, 7607–7632. URL: https://pubsonline.informs.org/doi/10.1287/mnsc.2022.4636. doi:10. 1287/mnsc.2022.4636.
- Arfaoui, N., Naeem, M. A., Maherzi, T., and Kayani, U. N. (2024). Can green investment funds hedge climate risk? Finance Research Letters, 60, 104961. URL: https://linkinghub.elsevier.com/retrieve/pii/S1544612323013338. doi:10.1016/j.frl.2023.104961.
- Baldauf, M., Garlappi, L., and Yannelis, C. (2020). Does climate change affect real estate prices? Only if you believe in it. The Review of Financial Studies, 33, 1256–1295. URL: https:

//academic.oup.com/rfs/article/33/3/1256/5735306. doi:10.1093/rfs/ hhz073.

- Baltas, K., Fiordelisi, F., and Mare, D. S. (2022). Alternative Finance after Natural Disasters. British Journal of Management, 33, 117–137. URL: https://onlinelibrary.wiley. com/doi/10.1111/1467-8551.12516. doi:10.1111/1467-8551.12516.
- Bansal, R., Kiku, D., and Ochoa, M. (2016). Price of Long-Run Temperature Shifts in Capital Markets. Working Paper 22529 National Bureau of Economic Research. URL: http://www.nber.org/papers/w22529.doi:10.3386/w22529.
- Batten, S., Sowerbutts, R., and Tanaka, M. (2016). Let's talk about the weather: The impact of climate change on Central Banks. Working paper 603 Bank of England. URL: https://www.ssrn.com/abstract=2783753. doi:10.2139/ssrn.2783753.
- Battiston, S., Dafermos, Y., and Monasterolo, I. (2021). Climate risks and financial stability. Journal of Financial Stability, 54, 100867. URL: https://linkinghub.elsevier.com/ retrieve/pii/S1572308921000267. doi:10.1016/j.jfs.2021.100867.
- Berg, F., Kölbel, J. F., and Rigobon, R. (2022). Aggregate confusion: The divergence of esg ratings. Review of Finance, 26. doi:10.1093/rof/rfac033.
- Bernstein, A., Gustafson, M. T., and Lewis, R. (2019). Disaster on the horizon: The price effect of sea level rise. Journal of Financial Economics, 134, 253–272. URL: https://linkinghub.elsevier.com/retrieve/pii/S0304405X19300807. doi:10.1016/j.jfineco.2019.03.013.
- Blasberg, A., Kiesel, R., and Taschini, L. (2022). Carbon Default Swap Disentangling the Exposure to Carbon Risk Through CDS. London School of Economics (LSE), Working paper., . doi:10.2139/ssrn.4257215.
- Boffo, R., Marshall, C., and Palatano, R. (2020). ESG investing: Environmental pillar scoring and reporting. Technical Report OECD Paris. URL: www.oecd.org/finance/ esg-investing-environmental-pillar-scoring-and-reporting.pdf.
- Bolton, P., and Kacperczyk, M. (2021). Do investors care about carbon risk? Journal of Financial Economics, 142, 517–549. doi:10.1016/j.jfineco.2021.05.008.
- Bolton, P., and Kacperczyk, M. T. (2023). Global Pricing of Carbon-Transition Risk. The Journal of Finance.
- Bonato, M., Cepni, O., Gupta, R., and Pierdzioch, C. (2022). Climate risks and realized volatility of major commodity currency exchange rates. Journal of Financial Markets, (p. 100760). URL: https://www.sciencedirect.com/science/article/ pii/S1386418122000519. doi:https://doi.org/10.1016/j.finmar.2022. 100760.

- Bouri, E., Dudda, T. L., Rognone, L., and Walther, T. (2023a). Climate Risk and the Nexus of Clean Energy and Technology Stocks. Annals of Operations Research. doi:10.1007/s10479-023-05487-z.
- Bouri, E., Iqbal, N., and Klein, T. (2022). Climate policy uncertainty and the price dynamics of green and brown energy stocks. Finance Research Letters, 47, 102740. URL: https://linkinghub.elsevier.com/retrieve/pii/S1544612322000629. doi:10.1016/j.frl.2022.102740.
- Bouri, E., Rognone, L., Sokhanvar, A., and Wang, Z. (2023b). From climate risk to the returns and volatility of energy assets and green bonds: A predictability analysis under various conditions. Technological Forecasting and Social Change, 194. doi:10.1016/j.techfore. 2023.122682.
- Breeden, D. T. (1979). An intertemporal asset pricing model with stochastic consumption and investment opportunities. Journal of Financial Economics, 7. doi:10.1016/0304-405X(79) 90016-3.
- Cepni, O., Demirer, R., Pham, L., and Rognone, L. (2023). Climate uncertainty and information transmissions across the conventional and ESG assets. Journal of International Financial Markets, Institutions and Money, 83, 101730. URL: https://www.sciencedirect. com/science/article/pii/S1042443122002025. doi:https://doi.org/10. 1016/j.intfin.2022.101730.
- Cepni, O., Demirer, R., and Rognone, L. (2022). Hedging climate risks with green assets. Economics Letters, 212, 110312. URL: https://linkinghub.elsevier.com/ retrieve/pii/S0165176522000222.doi:10.1016/j.econlet.2022.110312.
- Chatterji, A. K., Durand, R., Levine, D. I., and Touboul, S. (2016). Do ratings of firms converge? Implications for managers, investors and strategy researchers. Strategic Management Journal, 37, 1597–1614. URL: https://onlinelibrary.wiley.com/doi/10.1002/smj. 2407. doi:10.1002/smj.2407.
- Choi, D., Gao, Z., and Jiang, W. (2020). Attention to Global Warming. The Review of Financial Studies, 33, 1112–1145. URL: https://academic.oup.com/rfs/article/33/3/1112/5735304. doi:10.1093/rfs/hhz086.
- Cruz, J.-L., and Rossi-Hansberg, E. (2023). The economic geography of global warming. Review of Economic Studies, . doi:10.1093/restud/rdad042.
- Dafermos, Y., Gabor, D., Nikolaidi, M., Pawloff, A., and Van Lerven, F. (2020). Decarbonising is easy: Beyond market neutrality in the ECB's corporate QE. Technical Report New Economics Foundation.
- Dang, T. L., Moshirian, F., and Zhang, B. (2015). Commonality in news around the world. Journal of Financial Economics, 116, 82–110. URL: https://linkinghub.elsevier.com/ retrieve/pii/S0304405X14002530. doi:10.1016/j.jfineco.2014.11.007.

- Dechezleprêtre, A., Muckley, C. B., and Neelakantan, P. (2021). Is firm-level clean or dirty innovation valued more? The European Journal of Finance, 27, 31–61. URL: https://doi.org/ 10.1080/1351847X.2020.1785520. doi:10.1080/1351847X.2020.1785520. arXiv:https://doi.org/10.1080/1351847X.2020.1785520.
- Do, Q. A., Phan, V., and Nguyen, D. T. (2022). How do local banks respond to natural disasters? The European Journal of Finance, 0, 1–26. URL: https://doi.org/ 10.1080/1351847X.2022.2055969. doi:10.1080/1351847X.2022.2055969. arXiv:https://doi.org/10.1080/1351847X.2022.2055969.
- Drudi, F., Moench, E., Holthausen, C., Weber, P.-F., Ferrucci, G., Setzer, R., Nino, V. D., Barbiero, F., Faccia, D., Breitenfellner, A., Faiella, I., Farkas, M., maurice bun, Fornari, F., Ciccarelli, M., Paries, M. D., Giovannini, A., Papadopoulou, N., Parker, M., Petroulakis, F., Hebbink, G., Piloiu, A., Hlásková, S. J., Ploj, G., Popov, A. A., Kapp, D., Lopez-Garcia, P., Lozej, M., Lydon, R., Manninen, O., Roos, M., Manzanares, A., Santabárbara, D., Schotten, G., Mikkonen, K., Sotomayor, B., Stracca, L., Mongelli, F. P., Montes-Galdón, C., Müller, G., Nerlich, C., Osiewicz, M., Cruz, L. V., Page, A., Avgousti, A., Diez-Caballero, A., Aubrechtova, J., Mendoza, A. F., Téllez, M. D., Grüning, P., Garcia-Sanchez, P., Isgro, L., Bua, G., Kashama, M. K., Queiroz, R., Rognone, L., Röhe, O., McInerney, N., Meinerding, C., Tsalaporta, P., van den End, J. W., Theofilakou, A., Weth, M., Osorno-Torres, B., and Ouvrard, J.-F. (2021). Climate Change and Monetary Policy in the Euro Area. European Central Bank Occasional Paper Np, 2021271. doi:10.2139/ssrn.3928292.
- ECB (2021). Shining a light on climate risks: the ECB's economy-wide climate stress test. European Central Bank.
- ECB (2023). Financial Stability Review May 2023. The European Central Bank.
- Engle, R. F., Giglio, S., Kelly, B., Lee, H., and Stroebel, J. (2020). Hedging climate change news. The Review of Financial Studies, 33, 1184–1216. URL: https://academic.oup.com/ rfs/article/33/3/1184/5735305. doi:10.1093/rfs/hhz072.
- Eurostat (2008). NACE Rev. 2: Statistical classification of economic activities in the European Community.
- Faccini, R., Matin, R., and Skiadopoulos, G. (2023). Dissecting climate risks: Are they reflected in stock prices? Journal of Banking & Finance, 155, 106948. URL: https: //www.sciencedirect.com/science/article/pii/S037842662300153X. doi:https://doi.org/10.1016/j.jbankfin.2023.106948.
- Fama, E. F., and French, K. R. (2015). A five-factor asset pricing model. Journal of Financial Economics, 116, 1–22. URL: https://linkinghub.elsevier.com/retrieve/ pii/S0304405X14002323. doi:10.1016/j.jfineco.2014.10.010.
- Fava, S. D., Gupta, R., Pierdzioch, C., and Rognone, L. (2024). Forecasting international financial stress: The role of climate risks. Journal of International Financial Markets, Institutions and Money, 92, 101975. URL: https://linkinghub.elsevier.com/retrieve/pii/ S1042443124000416. doi:10.1016/j.intfin.2024.101975.

- Flori, A., Pammolli, F., and Spelta, A. (2021). Commodity prices co-movements and financial stability: A multidimensional visibility nexus with climate conditions. Journal of Financial Stability, 54, 100876. URL: https://linkinghub.elsevier.com/retrieve/ pii/S157230892100036X. doi:10.1016/j.jfs.2021.100876.
- Gentzkow, M., Kelly, B., and Taddy, M. (2019). Text as Data. Journal of Economic Literature, 57, 535–574. URL: https://pubs.aeaweb.org/doi/10.1257/jel.20181020. doi:10.1257/jel.20181020.
- Gibson Brandon, R., Krueger, P., and Schmidt, P. S. (2021). ESG Rating Disagreement and Stock Returns. Financial Analysts Journal, 77, 104–127. URL: https: //www.tandfonline.com/doi/full/10.1080/0015198X.2021.1963186. doi:10.1080/0015198X.2021.1963186.
- Giglio, S., Kelly, B., and Stroebel, J. (2021). Climate Finance. Annual Review of Financial Economics, 13. URL: https://www.annualreviews.org/doi/10.1146/annurev-financial-102620-103311. doi:10.1146/annurev-financial-102620-103311.
- Goldsmith-Pinkham, P., Gustafson, M. T., Lewis, R. C., and Schwert, M. (2023). Sea-Level Rise Exposure and Municipal Bond Yields. The Review of Financial Studies. doi:10.1093/rfs/hhad041.
- Goodell, J. W., Nammouri, H., Saâdaoui, F., and Jabeur, S. B. (2023). Carbon allowances amid climate change concerns: Fresh insights from wavelet multiscale analysis. Finance Research Letters, 55, 103871. doi:10.1016/J.FRL.2023.103871.
- Gupta, R., and Pierdzioch, C. (2023). Climate risk and the volatility of agricultural commodity price fluctuations: A prediction experiment. In D. Bourghelle, P. Grandin, F. Jawadi, and P. Rozin (Eds.), Behavioral Finance and Asset Prices: The Influence of Investor's Emotions (pp. 23–44). Cham: Springer International Publishing. URL: https://doi.org/10.1007/978-3-031-24486-5\_2. doi:10.1007/978-3-031-24486-5\_2.
- Görgen, M., Nerlinger, M., and Wilkens, M. (2020). Carbon Risk. Working paper 2930897 SSRN Electronic Journal. URL: http://www.ssrn.com/abstract=2930897.
- Hong, H., Karolyi, G. A., and Scheinkman, J. A. (2020). Climate Finance. The Review of Financial Studies, 33, 1011–1023. URL: https://academic.oup.com/rfs/article/33/3/ 1011/5735309. doi:10.1093/rfs/hhz146.
- Hong, H., Li, F. W., and Xu, J. (2019). Climate risks and market efficiency. Journal of Econometrics, 208, 265–281. URL: https://linkinghub.elsevier.com/retrieve/pii/ S0304407618301817. doi:10.1016/j.jeconom.2018.09.015.
- Hsu, P. H., Li, K., and Tsou, C. Y. (2023). The pollution premium. Journal of Finance, 78. doi:10.1111/jofi.13217.

- Huynh, T. D., and Xia, Y. (2021). Climate change news risk and corporate bond returns. Journal of Financial and Quantitative Analysis, 56, 1985–2009. URL: https: //www.cambridge.org/core/product/identifier/S0022109020000757/ type/journal{\_}article.doi:10.1017/S0022109020000757.
- In, S., Park, K., and Monk, A. (2019). Is "being green" rewarded in the market? An empirical investigation of decarbonization risk and stock returns. Working paper Stanford Global Projects Center.
- IPCC (2014). AR5 Climate change 2014: Impacts, adaptation, and vulnerability. Fifth Assessment Report Intergovernmental Panel on Climate Change.
- IPCC (2023). IPCC AR6 Synthesis Report: Climate Change 2023. Intergovernmental Panel on Climate Change.
- Karmakar, S., Gupta, R., Cepni, O., and Rognone, L. (2023). Climate risks and predictability of the trading volume of gold: Evidence from an ingarch model. Resources Policy, 82, 103438. doi:10.1016/J.RESOURPOL.2023.103438.
- Krueger, P., Sautner, Z., and Starks, L. T. (2020). The importance of climate risks for institutional investors. The Review of Financial Studies, 33, 1067–1111. URL: https://academic.oup.com/rfs/article/33/3/1067/5735302. doi:10.1093/rfs/hhz137.
- Kruttli, M. S., Tran, B. R., and Watugala, S. W. (2021). Pricing Poseidon: Extreme Weather Uncertainty and Firm Return Dynamics. Federal Reserve Bank of San Francisco, Working Paper Series (2021). doi:10.2139/ssrn.3451323.
- Meinerding, C., Schüler, Y. S., and Zhang, P. (2023). Shocks to Transition Risk. SSRN Electronic Journal. doi:10.2139/ssrn.4391260.
- Murfin, J., and Spiegel, M. (2020). Is the risk of sea level rise capitalized in residential real estate? Review of Financial Studies, 33, 1217–1255. doi:10.1093/rfs/hhz134.
- Newey, W. K., and West, K. D. (1987). A Simple, Positive Semi-Definite, Heteroskedasticity and Autocorrelation Consistent Covariance Matrix Author(s): Whitney K. Newey and Kenneth D. West Source:. Econometrica, 55, 703–708. URL: https://teach.business.uq.edu. au/courses/FINM6905/files/module-1/readings/NeweyWest.pdf.
- NGFS (2020). Guide for supervisors: Integrating climate-related and environmental risks into prudential supervision. Technical document Network for Greening the Financial System.
- Nordhaus, W. D., and Yang, Z. (1996). A regional dynamic general-equilibrium model of alternative climate-change strategies. American Economic Review, 86.
- Painter, M. (2020). An inconvenient cost: The effects of climate change on municipal bonds. Journal of Financial Economics, 135, 468–482. URL: https://linkinghub.elsevier. com/retrieve/pii/S0304405X19301631. doi:10.1016/j.jfineco.2019.06. 006.

- Pankratz, N., Bauer, R., and Derwall, J. (2023). Climate Change, Firm Performance, and Investor Surprises. Management Science. doi:10.1287/mnsc.2023.4685.
- Pástor, L., Stambaugh, R. F., and Taylor, L. A. (2021). Sustainable investing in equilibrium. Journal of Financial Economics, 142, 550–571. URL: https://linkinghub.elsevier.com/ retrieve/pii/S0304405X20303512. doi:10.1016/j.jfineco.2020.12.011.
- Pástor, L., Stambaugh, R. F., and Taylor, L. A. (2022). Dissecting green returns. Journal of Financial Economics, 146. doi:10.1016/j.jfineco.2022.07.007.
- Pietsch, A., and Salakhova, D. (2022). Pricing of Green Bonds: Drivers and Dynamics of the Greenium. European Central Bank (ECB), Working Paper Series No 2728 / September 2022. doi:10.2139/ssrn.4227559.
- Ramelli, S., Wagner, A. F., Zeckhauser, R. J., and Ziegler, A. (2021). Investor Rewards to Climate Responsibility: Stock-Price Responses to the Opposite Shocks of the 2016 and 2020 U.S. Elections. The Review of Corporate Finance Studies, 10, 748–787. URL: https://academic. oup.com/rcfs/article/10/4/748/6303621. doi:10.1093/rcfs/cfab010.
- Rognone, L., Hyde, S., and Zhang, S. S. (2020). News sentiment in the cryptocurrency market: An empirical comparison with Forex. International Review of Financial Analysis, 69, 101462. URL: https://linkinghub.elsevier.com/retrieve/pii/ S105752192030106X.doi:10.1016/j.irfa.2020.101462.
- Salisu, A. A., Olaniran, A., and Lasisi, L. (2023). Climate risk and gold. Resources Policy, 82, 103494. URL: https://www.sciencedirect.com/science/article/pii/ S0301420723002027. doi:https://doi.org/10.1016/j.resourpol.2023. 103494.
- Schmitz, S., Dawson, B., Spannagle, M., Thomson, F., Koch, J., and Eaton, R. (2004). The Greenhouse Gas Protocol - A Corporate Accounting and Reporting Standard, Revised Edition. The GHG Protocol Corporate Accounting and Reporting Standard, 9, 116. URL: http://www.ghgprotocol.org/files/ghg-protocol-revised. pdf.arXiv:1-56973-568-9.
- Sena, V., Kanungo, Ozdemir, S., Yannopoulou, and Patel, P. R. P., N., (2022).Are Reshoring Decisions Influenced by Stakeholders and External Country-Level Environmental Regulation?. British Journal of Management. URL: https://onlinelibrary.wiley.com/doi/abs/10.1111/ doi:https://doi.org/10.1111/1467-8551.12680. 1467-8551.12680. arXiv:https://onlinelibrary.wiley.com/doi/pdf/10.1111/1467-8551.12680.
- Stern, N. (2007). The Economics of Climate Change volume 9780521877. Cambridge University Press. URL: https://www.cambridge.org/core/product/identifier/ 9780511817434/type/book. doi:10.1017/CB09780511817434.

Date PKI	Physical risk news tonics	Physical risk relevant news titles
		A AND USE A LIVE A LIVE A LIVE A LIVE A LIVE A LIVE A
19/09/2018 12.3% Ar Oc	tic Sea Minimum; Melting glaciers; Permafrost thawing sean acidification; Sea level rise	Greenland and the hunt for better climate science; In Greenland a glacier's collapse shows climate impact; Coceral cuts sharply EU cereal rapeseed crop estimates
11/02/2007 9.8% <sup>IP</sup>	CC draft report by the U.N. climate panel in Paris; Climate ange affects fishery; Sea level rise	Global warming impacts of temperature rises; Dratt indings by U.N. climate panel; Seas rising faster than U.N. predicts study, U.N. panel to link warming to humans project more; Cool water surges could affect fish stocks report; U.N. panel says "very likely" humans cause warming
6/12/2019 9.6% <sup>Bi</sup>	odiversity Loss; Europe must protect rivers and lakes; shing, reef, tourism at risk; Increased costs due to climate stress	Europe must do more to protect its rivers and lakes: scientists; Runaway warming could sink fishing and reef tourism researchers warn; Forward prices slip on wetter weather view lower German rates
7/03/2012 9.1% GI	obal warming irreversible; Spain droughts, water stress; ost impact on agriculture	Global warming close to becoming irreversible scientists; Spain drought hits hydro, irrigation stocks again; EU wheat too expensive for export - Toepfer
0/01/2007 8.9% Hu An	unger due to water scarcity; Adaptation measure for companies; iimal migration	Clouds a puzzle for U.N. global warming panel; Millions to go hungry, waterless; Bird ranges move, but is it climate change?; Europe's wheat crop on track, frost fears ease; Commulsorv water meterine takes sten closer
7/02/2020 8.7% Go	vernments face pressure to protect biodiversity	Governments face pressure to protect nature in biodiversity
8/09/2006 8.7% Ra	ising temperatures causes floods, droughts, erosion; Hungary k sand dunes, floods; Rising water costs; Water scarcity	Eco-paradises in crossfire of water scarcity fight; Hungary region battles advancing sand dunes, floods; Water everywhere but not clean enough to drink; Investors bet on rising costs for scarce water. Stanish wetland struooles as water levels dron
5/04/2012 8.6% Per	ople cause climate change, need to understand impact on humans	s Climate contrarian case wilts: Gerard Wynn
4/12/2009 8.5% Fo	rests to climate fight; Coral climate crisis costs; Biodiversity loss; litic engagement in climate fight	Climate change makes finding Nemo even harder report; Coral climate crisis puts 250 million <sup>34</sup> at risk: U.N.; Forest communities said key to climate fight; Natural disasters at decade low in 2009-UN report; Antarctic researcher commutes across continents for work
3/08/2019 8.4% Fo.	od security	Farming and eating need to change to curb global warming: UN report
Date TRI	Transition risk news topics	Transition risk relevant news titles
1/08/2011 19.1% EL	J GHG emissions higher than reported	EU HFC emissions higher than reported; New research links cosmic rays to cloud formation
KN %C.11 1002/80/9	oto Protocol; Certified Emission Reduction (CER)	Kyoto projects harm ozone layer: U.N. official; Daily secondary CEK market report
1/09/2010 16.3% Ky	oto CO2 scheme probed; CER	HFC cutting plants under Kyoto CO2 scheme probed; EU carbon permit volumes fall in Aug, CERs rise
$5/09/2010$ 15.0% $\frac{\text{Re}}{\text{ind}}$	view of carbon offsets; Low-carbon heating sources raises	UN to review CO2 offset request from 9th plant; UK business group warns on new low-carbon
5/10/2015 13.8% En	erect years nissions regulation; Increase in energy costs	ourpout U.S. announces new moves to limit super greenhouse gases; German 2016 green power surcharge at 6.354 cents/kWh - grid firms
9/08/2011 13.8% Co	stly transition	Money spinning China carbon scheme may end with loss; Sberbank CO2 role questioned after hune issumme
006/2010 12.5% CE	3R issuances at lowest; Costly transition; International carbon	Nuclear instants. The second of the second o
am and a second s	urket	costly with tiny benefits; EU carbon hits fresh 3-wk low on weak German power
4/08/2012 11.8% CE	eks request	Developers seek 4./ mln UERs, incl. 1.5 mln HFC units Nineteen EU nations back common position on carbon market reform: France's EDF hydropower
8/02/2017 11.4% Ca	rbon Reform Deal; Renewable energies; Emissions targets	availability down 1.3 GW due to strike; Marshall Islands first to ratify global HFC greenhouse gas pact; Austria's EVN puts Bulgarian hydropower project on hold
6/09/2010 11.4% Gr and	een incentive for coal; Clean energy projects; Carbon Capture d Storage technologies	UN panel to rule on green incentives for coal; Ozone recovering but will take longer over poles; UN gives CO2 auditors time to study liability plan; Carbon capturing technology doomed in Europe - study; Mexico says world should trust U.S. on emissions

Table 1: Physical and transition risk top news articles

days Note: This table reports the dates, the Physical Risk Index (PRI) and Transition Kisk Index (1KI), the main news topics, and use or revenue with the highest physical and transition risk over the period Jan 2005-Oct 2021. News sourced from Reuters News with a European regional focus.

	Concern, pp v 100	Concern, T. v. 100
	$\operatorname{Concern}_{t,PR} \ge 100$	$\operatorname{Concern}_{t,TR} \times 100$
Drift $c$	7.863	8.462
	(0.047)	(0.061)
$\phi$	0.326	0.413
	(0.014)	(0.014)

Table 2: AR1 estimates of physical risk and transition risk concern

Note: Estimates of autoregressive process of order 1 (AR1) concern time series on physical risk, as in Equation (1a), and transition risk as in Equation (1b) for the period Jan 2005-Oct 2021. Standard error in parenthesis.

Panel a) Numbe	r of firms wit	h data		
Year	$log-GHG_E$	$\log$ -GHG <sub>EI</sub>	E score	ESG score
2005	259	254	537	686
2006	341	336	584	717
2007	423	413	706	777
2008	463	445	756	805
2009	547	518	776	823
2010	587	548	817	852
2011	626	583	833	874
2012	659	608	859	897
2013	689	634	880	913
2014	734	670	900	939
2015	773	707	945	978
2016	811	732	962	994
2017	857	767	1,011	1,044
2018	907	797	1,056	1,089
2019	944	818	1,074	1,105
2020	953	811	1,087	1,115
2021	949	793	1,081	1,116
Panel b) Thresh	old			
25th Percentile	4.45	0.97	34.55	38.15
75th Percentile	6.00	2.14	76.46	69.56

 Table 3: GHG emissions level and intensity, E score, and ESG score data

Note: Exposure data as number of firm with data coverage over time (from 2005 to 2021) for the EuroStoxx 600 Index constituents (panel a) and the threshold levels to construct green and brown portfolios (panel b) for log-GHG emissions levels (GHG<sub>E</sub>), log-GHG emissions intensity (GHG<sub>EI</sub>), Environmental score (E score), and Environmental, Social, and Governance score (ESG score). Environmental, ESG, and GHG emissions data are sourced from Refinitiv.

<ul> <li>D Electricity, gas, steam and air</li> <li>B Mining and quarrying</li> <li>M Professional, scientific and tec</li> <li>H Transportation and storage</li> <li>C Manufacturing</li> <li>N Administrative and support se</li> <li>Water supply; sewerage, wast</li> <li>F Construction</li> <li>G Wholesale and retail trade; ref</li> <li>I Accommodation and food servities</li> <li>K Financial and insurance activities</li> <li>S Other service activities</li> </ul>		No.	$\log_{\rm GHG_{\it E}}$	$\log_{GHG_{EI}}$	ESG	Щ	$\operatorname{GHG}_E$ c Index	ontribution EU
<ul> <li>B Mining and quarrying</li> <li>M Professional, scientific and tec</li> <li>H Transportation and storage</li> <li>C Manufacturing</li> <li>N Administrative and support se</li> <li>Water supply; sewerage, waste</li> <li>F Construction</li> <li>G Wholesale and retail trade; rep</li> <li>I Accommodation and food ser</li> <li>K Financial and insurance activi</li> <li>J Information and communicati</li> <li>A Agriculture, forestry and fishii</li> <li>S Other service activities</li> </ul>	supply	37	17.09	7.77	56.73	60.4	41%	28%
<ul> <li>M Professional, scientific and tec</li> <li>H Transportation and storage</li> <li>C Manufacturing</li> <li>N Administrative and support se</li> <li>N Mater supply; sewerage, waste</li> <li>F Construction</li> <li>G Wholesale and retail trade; rej</li> <li>Accommodation and food ser</li> <li>K Financial and insurance activities</li> <li>S Other service activities</li> </ul>		48	16.52	6.95	59.27	59.29	23%	2%
<ul> <li>H Transportation and storage</li> <li>C Manufacturing</li> <li>N Administrative and support se</li> <li>E Water supply; sewerage, waste</li> <li>F Construction</li> <li>G Wholesale and retail trade; rep</li> <li>I Accommodation and food serving</li> <li>K Financial and insurance activit</li> <li>J Information and communication</li> <li>A Agriculture, forestry and fishin</li> <li>S Other service activities</li> </ul>	ies	150	15.62	7.64	51.74	48.04	9%6	1%
C Manufacturing N Administrative and support se E Water supply; sewerage, waste F Construction G Wholesale and retail trade; rep I Accommodation and food ser K Financial and insurance activi J Information and communicati A Agriculture, forestry and fishi S Other service activities		37	15.36	6.16	55.75	55.93	7%	14%
<ul> <li>N Administrative and support se</li> <li>E Water supply; sewerage, waste</li> <li>F Construction</li> <li>G Wholesale and retail trade; rej</li> <li>Accommodation and food sert</li> <li>K Financial and insurance activi</li> <li>J Information and communicati</li> <li>A Agriculture, forestry and fishin</li> <li>S Other service activities</li> </ul>		309	15.36	6.09	56.15	53.57	7%	26%
<ul> <li>E Water supply; sewerage, waste</li> <li>F Construction</li> <li>G Wholesale and retail trade; rej</li> <li>G Wholesale and retail trade; rej</li> <li>I Accommodation and food servic</li> <li>K Financial and insurance activities</li> <li>S Other service activities</li> </ul>	SS	36	14.87	4.86	47.65	40.75	4%	1%
<ul> <li>F Construction</li> <li>G Wholesale and retail trade; rep</li> <li>I Accommodation and food serving</li> <li>K Financial and insurance activities</li> <li>A Agriculture, forestry and fishing</li> <li>S Other service activities</li> </ul>	t and remediation activities	6	14.43	6.08	61.2	61.19	3%	5%
<ul> <li>G Wholesale and retail trade; rej</li> <li>I Accommodation and food ser</li> <li>K Financial and insurance activi</li> <li>J Information and communicati</li> <li>A Agriculture, forestry and fishin</li> <li>S Other service activities</li> </ul>		35	13.84	4.76	58.88	65.16	2%	2%
<ul> <li>I Accommodation and food ser</li> <li>K Financial and insurance activi</li> <li>J Information and communicati</li> <li>A Agriculture, forestry and fishii</li> <li>S Other service activities</li> </ul>	vehicles and motorcycles	81	13.66	4.83	51.48	47.81	1%	3%
<ul><li>K Financial and insurance activi</li><li>J Information and communicati</li><li>A Agriculture, forestry and fishii</li><li>S Other service activities</li></ul>		16	13.18	5.71	57.7	55.2	1%	1%
<ul> <li>J Information and communicati</li> <li>A Agriculture, forestry and fishi</li> <li>S Other service activities</li> </ul>		252	12.67	3.84	50.67	53.52	0%0	0%
A Agriculture, forestry and fishi S Other service activities		109	12.53	3.36	50.74	41.76	0%0	0%0
S Other service activities			12.12	4.07	54.92	39.16	0%0	16%
		9	11.81	4.42	45.04	42.05	0%0	0%0
Q Human health and social work		6	11.67	4.21	47.1	39.97	0%0	1%
O Public administration and defe	sory social security	4	11.19	3.08	41.36	41.53	0%0	1%
R Arts, entertainment and recrea		12	10.67	3.55	47.08	39.91	0%0	0%0
L Real estate activities		47	10.38	4.30	48.02	49.79	0%0	0%0

Table 4: EuroStoxx 600 Index historical constituents' sectoral composition

sector (No.), average environmental score (E score), environmental, social, and governance score (ESG score), log-GHG emissions levels (log GHG<sub>E</sub>), log-GHG emissions intensity (log GHG<sub>EI</sub>). Per-year average GHG emission contribution of each EuroStoxx 600 Index NACE sector to the total EuroStoxx 600 Index emissions (GHG $_E$  contribution Index), and the per-year average GHG emissions of the full NACE sector to the total European Union GHG emission (GHG $_E$ the sector, the darker the colour, the 'browner' the sector according to each metric (E score, ESG score, GHG<sub>EI</sub>, GHG<sub>E</sub>). Environmental, ESG, and GHG emissions Note: EuroStoxx 600 Index historical constituents sectoral (NACE code - sector) composition over the period Jan 2005-Oct 2021, number of companies per contribution EU) as from EU27 sourced from Eurostat. The table is sorted according to descending greenhouse gas emissions. The lighter the colour the 'greener' data are sourced from Refinitiv.

omposition
portfolios co
$\operatorname{GHG}_E$
$GHG_{EI}$ ,
ESG,
щ
d brown
an
Green
Table 5:

Pan	el a)	E sc	ore	ESG	score	log-GH	$G_{EI}$	log-GH	$\Im_E$
		Green	Brown	Green	Brown	Green	Brown	Green	Brown
Mei	ric average	85.64	19.89	78.77	25.42	0.92	6.47	8.69	15.67
Nui	nber of assets	456	708	524	969	396	285	409	241
Pan	el b) Sectoral composition (%)								
NA	CE code - sector								
A	Agriculture, forestry and fishing	0.00	0.14	0.19	0.14	0.00	0.00	0.00	0.00
В	Mining and quarrying	4.82	3.95	4.58	3.45	1.01	12.63	1.47	11.62
U	Manufacturing	28.73	26.55	30.73	27.16	14.65	33.68	16.87	34.44
Ω	Electricity, gas, steam and air conditioning supply	4.17	2.97	3.82	2.59	0.51	9.82	0.98	11.62
Щ	Water supply; sewerage; waste management and remediation activities	0.88	0.56	0.95	0.29	0.00	2.11	0.24	2.07
Ľ	Construction	5.48	2.97	3.82	2.73	3.28	3.51	3.42	4.15
IJ	Wholesale and retail trade; repair of motor vehicles and motorcycles	5.26	7.91	5.53	6.61	5.05	2.81	5.87	6.64
Η	Transporting and storage	3.51	2.40	3.44	2.87	1.26	7.02	1.71	7.05
I	Accommodation and food service activities	1.75	1.69	1.53	1.29	1.26	1.40	0.98	0.83
J	Information and communication	5.48	9.75	7.82	8.91	13.89	2.11	12.71	2.90
$\mathbf{N}$	Financial and insurance activities	24.12	19.63	18.32	20.98	39.65	4.56	33.50	5.39
Γ	Real estate activities	3.29	3.25	3.24	4.17	3.79	3.51	7.33	0.00
Σ	Professional, scientific and technical activities	9.87	12.15	12.98	11.78	10.10	13.68	9.54	11.62
Z	Administrative and support service activities	1.75	2.82	1.53	3.30	3.28	1.75	2.93	1.66
0	Public administration and defence; compulsory social security	0.22	0.56	0.19	0.57	0.51	0.00	0.49	0.00
0	Human health and social work activities	0.00	0.56	0.38	0.72	0.25	0.35	0.24	0.00
Ч	Arts, entertainment and recreation	0.22	1.41	0.57	1.58	1.26	0.35	1.47	0.00
S	Other services activities	0.44	0.71	0.38	0.86	0.25	0.70	0.24	0.00
te: W	s sort the EuroStoxx 600 Index historical constituents on environmental score ()	E score), e	environme	ental, soci	al and gov	ernance s	core (ESG	i score), g	reenhouse g
iission	$\alpha$ intensity (GHG $_{ET}$ ), preenhouse gas emissions (GHG $_{E}$ ) and we create individu	al preen	and brown	nortfolio	s This tal	ale renorts	each nort	folio meti	ic average (F

Note: We sort the EuroStoxx 600 Index historical constituents on environmental score (E score), environmental, social and governance score (ESG score), greenhouse gas emissions intensity (GHG<sub>EI</sub>), greenhouse gas emissions (GHG<sub>E</sub>) and we create individual green and brown portfolios. This table reports each portfolio metric average (E, ESG, GHG<sub>EI</sub>), the number of assets, and the relative NACE sectoral (%) composition over the period Jan 2005-Oct 2021. Environmental, ESG, and GHG emissions data are sourced from Refinitiv.

Panel a) Transition	lisk bete	i portioi	105						
Full sample TRI	(	Quintile	s		Deciles		25	Percent	iles
	L	Н	LMH	L	Н	LMH	L	Н	LMH
E(R)-Rf (%)	11.93	12.53	-1.17	10.78	10.19	-0.10	8.46	5.71	1.96
$\sigma$ (%)	18.97	19.35	5.78	20.11	20.58	8.25	22.35	22.47	13.05
SR	0.63	0.65	-0.20	0.54	0.50	-0.01	0.38	0.25	0.15
Before 2015 TRI	(	Quintile	s		Deciles		25	Percent	iles
	L	Н	LMH	L	Н	LMH	L	Н	LMH
E(R)-Rf (%)	10.21	12.99	-3.81	9.40	11.66	-3.38	8.32	10.21	-3.08
$\sigma$ (%)	19.77	20.28	5.94	20.97	21.40	8.06	23.53	23.16	12.76
SR	0.52	0.64	-0.64	0.45	0.54	-0.42	0.35	0.44	-0.24
After 2015 TRI	(	Quintile	s		Deciles		25	Percent	iles
	L	Н	LMH	L	Н	LMH	L	Н	LMH
E(R)-Rf (%)	14.53	11.97	2.74	12.88	8.18	4.81	8.68	-0.41	9.61
$\sigma$ (%)	17.77	18.16	5.95	18.83	19.37	8.51	20.57	21.46	13.44
SR	0.82	0.66	0.46	0.68	0.42	0.57	0.42	-0.02	0.71
Panel b) Physical ris	sk beta j	ortfolio	DS						
Full sample PRI	(	Quintile	s		Deciles		25	Percent	iles
	L	Н	LMH	L	Н	LMH	L	Н	LMH
E(R)-Rf (%)	11 78	10.01	1 00	10.00	10 50	0.00	7 00	5 04	0.43
	11.70	13.31	-1.98	10.98	10.58	-0.28	7.08	5.94	0.45
$\sigma$ (%)	19.23	13.31 19.35	-1.98 5.78	10.98 20.37	10.58 20.51	-0.28 8.00	7.08 22.83	5.94 23.07	13.29
σ (%) SR	19.23 0.61	13.31 19.35 0.69	-1.98 5.78 -0.34	10.98 20.37 0.54	10.58 20.51 0.52	-0.28 8.00 -0.04	7.08 22.83 0.31	5.94 23.07 0.26	13.29 0.03
σ (%) SR Before 2015 PRI	19.23 0.61	13.31 19.35 0.69 Quintile	-1.98 5.78 -0.34	10.98 20.37 0.54	10.58 20.51 0.52 Deciles	-0.28 8.00 -0.04	7.08 22.83 0.31 25	5.94 23.07 0.26 Percent	13.29 0.03 iles
σ (%) SR Before 2015 PRI	11.78 19.23 0.61 L	13.31 19.35 0.69 Quintiles H	-1.98 5.78 -0.34 s LMH	10.98 20.37 0.54	10.58 20.51 0.52 Deciles H	-0.28 8.00 -0.04	7.08 22.83 0.31 <u>25</u> L	3.94 23.07 0.26 Percent: H	13.29 0.03 iles LMH
σ (%) SR Before 2015 PRI E(R)-Rf (%)	19.23 0.61 L 11.66	13.31 19.35 0.69 Quintile H 12.82	-1.98 5.78 -0.34 s LMH -2.40	10.98 20.37 0.54 L 10.22	10.58 20.51 0.52 Deciles H 11.60	-0.28 8.00 -0.04 LMH -2.61	$     \begin{array}{r}       7.08 \\       22.83 \\       0.31 \\       \hline             \underline{25} \\             \overline{L} \\             6.11 \\             \end{array}     $	3.94 23.07 0.26 Percent H 8.71	13.29 0.03 iles LMH -3.75
$\sigma (\%)$ SR Before 2015 PRI $E(R)-Rf (\%)$ $\sigma (\%)$	11.76 19.23 0.61 L 11.66 20.16	13.31 19.35 0.69 Quintiles H 12.82 20.11	-1.98 5.78 -0.34 s LMH -2.40 5.75	10.98 20.37 0.54 L 10.22 21.43	10.58 20.51 0.52 Deciles H 11.60 21.26	-0.28 8.00 -0.04 LMH -2.61 7.86	7.08 22.83 0.31 25 L 6.11 23.71	3.94 23.07 0.26 Percent H 8.71 23.62	13.29 0.03 iles LMH -3.75 12.91
$\sigma (\%)$ SR Before 2015 PRI $E(R)-Rf (\%)$ $\sigma (\%)$ SR	11.76 19.23 0.61 L 11.66 20.16 0.58	13.31 19.35 0.69 Quintile: H 12.82 20.11 0.64	-1.98 5.78 -0.34 s LMH -2.40 5.75 -0.42	10.98 20.37 0.54 L 10.22 21.43 0.48	10.58 20.51 0.52 Deciles H 11.60 21.26 0.55	-0.28 8.00 -0.04 LMH -2.61 7.86 -0.33	7.08 22.83 0.31 25 L 6.11 23.71 0.26	3.94 23.07 0.26 Percent H 8.71 23.62 0.37	13.29 0.03 iles LMH -3.75 12.91 -0.29
σ (%) SR Before 2015 PRI E(R)-Rf (%) σ (%) SR After 2015 PRI	19.23 0.61 L 11.66 20.16 0.58	13.31 19.35 0.69 Quintile: H 12.82 20.11 0.64 Quintile:	-1.98 5.78 -0.34 s LMH -2.40 5.75 -0.42 s	10.98 20.37 0.54 L 10.22 21.43 0.48	10.58 20.51 0.52 Deciles H 11.60 21.26 0.55 Deciles	-0.28 8.00 -0.04 LMH -2.61 7.86 -0.33	7.08 22.83 0.31 25 L 6.11 23.71 0.26 25	3.94 23.07 0.26 Percent H 8.71 23.62 0.37 Percent	13.29 0.03 iles LMH -3.75 12.91 -0.29 iles
σ (%) SR Before 2015 PRI E(R)-Rf (%) σ (%) SR After 2015 PRI	11.78 19.23 0.61 L 11.66 20.16 0.58 C L	13.31 19.35 0.69 Quintiles H 12.82 20.11 0.64 Quintile H	-1.98 5.78 -0.34 s LMH -2.40 5.75 -0.42 s LMH	10.98 20.37 0.54 L 10.22 21.43 0.48 L	10.58 20.51 0.52 Deciles H 11.60 21.26 0.55 Deciles H	-0.28 8.00 -0.04 LMH -2.61 7.86 -0.33 LMH	7.08 22.83 0.31 25 L 6.11 23.71 0.26 25 L	3.94 23.07 0.26 Percent H 8.71 23.62 0.37 Percent H	13.29 0.03 iles LMH -3.75 12.91 -0.29 iles LMH
σ (%) SR Before 2015 PRI E(R)-Rf (%) σ (%) SR After 2015 PRI E(R)-Rf (%)	11.78 19.23 0.61 L 11.66 20.16 0.58 C L 12.14	13.31 19.35 0.69 Quintile: H 12.82 20.11 0.64 Quintile: H 14.11	-1.98 5.78 -0.34 s LMH -2.40 5.75 -0.42 s LMH -1.29	10.98         20.37         0.54         L         10.22         21.43         0.48         L         12.33	10.58 20.51 0.52 Deciles H 11.60 21.26 0.55 Deciles H 9.26	-0.28 8.00 -0.04 LMH -2.61 7.86 -0.33 LMH 3.26	$ \begin{array}{r} 7.08 \\ 22.83 \\ 0.31 \\ \hline                                   $	3.94 23.07 0.26 Percent H 8.71 23.62 0.37 Percent H 2.10	0.43 13.29 0.03 iles LMH -3.75 12.91 -0.29 iles LMH 6.71
σ (%) SR Before 2015 PRI E(R)-Rf (%) σ (%) SR After 2015 PRI E(R)-Rf (%) σ (%)	11.78 19.23 0.61 11.66 20.16 0.58 L 12.14 17.84	13.31 19.35 0.69 Quintile: H 12.82 20.11 0.64 Quintile: H 14.11 18.23	-1.98 5.78 -0.34 s LMH -2.40 5.75 -0.42 s LMH -1.29 5.83	10.98         20.37         0.54         L         10.22         21.43         0.48         L         12.33         18.77	10.58 20.51 0.52 Deciles H 11.60 21.26 0.55 Deciles H 9.26 19.42	-0.28 8.00 -0.04 LMH -2.61 7.86 -0.33 LMH 3.26 8.19	7.08 22.83 0.31 25 L 6.11 23.71 0.26 25 L 8.47 21.51	23.07 0.26 Percent H 8.71 23.62 0.37 Percent H 2.10 22.26	0.43 13.29 0.03 iles LMH -3.75 12.91 -0.29 iles LMH 6.71 13.80

 Table 6: Portfolios sorted on transition and physical climate betas

Note: This table shows the performances of the 5, 10, 25 low (L) and high (H) portfolios sorted according to their sensitivity to the Transition Risk Index (TRI) and Physical Risk Index (PRI), alongside the low-minus-high (LMH) transition and physical risk spread returns portfolios. We report the portfolios percentage annualised excess returns(E(R)-Rf), standard deviations ( $\sigma$ ), and the Sharpe ratios (SR), for three periods (full sample, Jan 2005-Oct 2021; before 2015, Jan 2005-Dec 2014; and after 2015, Jan 2015-Oct 2021) considering the EuroStoxx 600 Index constituents.

LMH -2.26	L 8.07	H 7.96	LMH -0.54	5.14 5.14	H	-0.07	arm sampro arFF5	8.46	9.20	LMH -1.32	L 1.67	H 7.46	LMH -0.45	L L 6.10		1.42 1.42
-2.26	8.07	7.96	-0.54	5.14	151	-0.07	$lpha_{FF5}$	8.46	9.20	-1.32	7.67	7.46	-0.45 0.72	د 6.10	3.94	1.42 1.42 0.46
				071	1	000	LT C			0.01			<i>cc</i> 0	0110		046
-	4	3.88	-0.27	1.0Y	1.65	70.0-		96.0	6.11		4.23	3.84	-11.44	2.19	1.51	Ì
0.00	101	1.04	-0.01	1.08	1.09	0.00	MKT≁	1.00	1.01	-0.01	1.03	1.04	-0.02	1.06	1.07	-0.02
0.07	68.77	73.09	-0.37	48.63	62.59	-0.18	[H]	111.53	65.17	-0.84	90.10	52.62	-0.86	63.70	44.24	-0.73
-0.01	0.50	0.52	-0.02	0.63	0.66	-0.03	SMB+	0.43	0.42	0.01	0.48	0.50	-0.02	0.54	0.58	-0.04
-0.20	15.96	15.14	-0.71	12.67	16.92	-0.68	[t]	18.46	13.78	0.36	15.41	12.17	-0.43	13.93	12.00	-0.71
-0.03	0.17	0.19	-0.02	0.23	0.25	-0.02	HML <sup>4</sup>	0.11	0.15	-0.04	0.15	0.18	-0.03	0.21	0.21	0.00
-0.90	6.15	4.78	-0.50	5.50	5.50	-0.49	[t]	5.84	5.59	-1.43	6.31	5.38	-1.00	6.91	5.54	-0.06
0.02	0.12	0.10	0.02	0.11	0.09	0.02	$CMA_{t}$	0.11	0.11	0.00	0.12	0.11	0.01	0.12	0.08	0.04
0.48	4.19	3.27	0.61	2.40	2.32	0.39	[t]	5.09	4.53	-0.03	4.47	3.52	0.28	3.05	1.99	0.72
-0.03	-0.15	-0.14	-0.02	-0.19	-0.23	0.04	$RMW_{t}$	-0.13	-0.10	-0.03	-0.16	-0.13	-0.02	-0.20	-0.20	0.00
-1.10	-8.04	-5.55	-0.54	-6.33	-7.01	1.16	[t]	-8.59	-5.30	-1.59	-8.23	-5.82	-0.86	-7.50	-6.03	-0.08
S		Deciles		25	Percen	iles	Before 2015		Juintiles			Deciles		25 I	Percentil	SS
TMH		H	LMH		H	LMH			H	LMH		H	LMH		H	LMH
-2.45	7.06	8.35	-2.57	3.15	6.06	-4.09	QUEFE	6.82	9.55	-3.85	6.05	8.33	-3.46	5.22	6.98	-3.01
-1.36	2.85	3.05	-1.00	0.85	1.64	-0.98	[t]	3.39	4.94	-2.14	2.37	3.41	-1.35	1.36	2.06	0.77
0.01	1.06	1.04	0.01	1.11	1.09	0.01	$MKT_{t}$	1.01	1.01	-0.01	1.04	1.05	-0.01	1.10	1.09	0.00
0.67	65.73	65.50	0.56	47.64	57.01	0.51	[t]	73.88	59.53	-0.39	61.41	54.77	-0.43	49.70	52.10	0.06
0.02	0 47	0.46	0.01	0.57	0.57	00.0	SMB.	0 43	0.38	0.05	0.46	0 44	0.02	0.51	0.51	000
0.74	13.11	11.66	0.38	10.88	12.01	0.02	E	11.39	12.03	1.53	9.05	11.43	0.42	7.19	11.80	0.02
-0.05	0.15	0.20	-0.05	0.19	0.25	-0.06	HMI #	0.10	0.16	-0.06	0.12	0.19	-0.06	0.16	0.17	-0.01
-1.15	4.17	3.93	-0.99	3.57	3.90	-0.88	[t]	3.34	4.79	-1.87	3.29	4.79	-1.81	3.24	3.81	-0.12
0.00	0.13	0.13	0.00	0.13	0.13	0.01	$CMA_{t}$	0.10	0.15	-0.05	0.12	0.16	-0.04	0.12	0.15	-0.04
0.03	3.46	4.01	-0.04	2.54	2.74	0.11	[t]	3.32	5.06	-1.24	3.21	4.87	-1.01	2.13	3.42	-0.51
-0.06	-0.21	-0.15	-0.06	-0.25	-0.25	0.00	$RMW_t$	-0.16	-0.12	-0.04	-0.19	-0.17	-0.02	-0.23	-0.24	-0.24
-1.79	-8.83	-4.19	-1.82	-6.95	-5.55	-0.07	[t]	-7.28	-4.95	-1.45	-6.64	-6.18	-0.72	-6.37	-6.38	0.04
s		Deciles		25	Percen	iles	After 2015		Duintiles			Deciles		25 I	Percentil	ss
LMH	Г	Н	LMH	Г	Н	LMH		Г	H	LMH	Г	Н	LMH	Г	Н	LMH
-2.13	10.51	8.54	2.27	9.85	3.95	6.14	$\alpha_{FF5}$	11.14	9.86	1.61	10.75	7.79	3.20	8.63	1.92	7.05
-0.83	3.80	3.04	0.68	2.15	0.97	1.15	[t]	6.00	4.40	0.69	4.24	2.64	1.01	2.10	0.49	1.41
-0.01	0.99	1.02	-0.03	1.02	1.05	-0.03	$MKT_t$	0.98	0.99	-0.02	0.99	1.02	-0.02	0.98	1.01	-0.04
-0.78	56.91	51.15	-1.71	35.85	41.03	-1.19	[t]	102.34	42.22	-0.57	86.54	33.29	-0.65	47.75	27.23	-1.15
-0.04	0.57	0.64	-0.07	0.76	0.82	-0.06	$SMB_t$	0.46	0.51	-0.05	0.54	0.61	-0.08	0.64	0.70	-0.06
-1.32	19.28	20.02	-1.52	14.92	21.77	-1.08	[t]	25.73	16.30	-1.45	22.46	13.20	-1.37	17.56	11.20	-0.85
0.01	0.17	0.13	0.04	0.24	0.21	0.03	$HML_t$	0.12	0.10	0.02	0.18	0.13	0.05	0.25	0.22	0.03
0.22	6.54	3.70	0.80	5.89	5.27	0.59	[t]	7.40	3.88	0.71	6.84	3.56	1.14	5.69	5.08	0.50
0.04	0.13	0.06	0.07	0.11	0.06	0.05	$CMA_t$	0.13	0.06	0.07	0.14	0.04	0.11	0.16	0.02	0.14
1.04	4.22	1.24	1.15	2.00	0.99	0.64	[t]	5.80	2.15	1.92	3.72	0.87	1.76	2.66	0.27	1.99
0.03	-0.07	-0.13	0.07	-0.09	-0.22	0.13	$RMW_t$	-0.08	-0.07	-0.02	-0.11	-0.09	-0.02	-0.14	-0.15	0.01
1.10	-2.89	-4.87	1.95	-2.45	-4.99	2.53	[1]	-4.43	-3.32	-0.55	-3.90	-3.01	-0.39	-3.66	-3.20	0.15
							2									
	-0.03 -0.09 -0.09 -0.03 -0.03 -0.03 -0.01 -1.10 -0.01 -0.05 -0.01 -0.05 -0.01 -0.05 -0.01 -0.05 -0.01 -0.05 -0.01 -0.05 -0.01 -0.03	$\begin{array}{c cccccc} -0.00 & 0.17 \\ -0.03 & 0.15 \\ 0.48 & 4.19 \\ -0.03 & -0.15 \\ -1.10 & -8.04 \\ 8 & -1.36 & -2.45 \\ -1.36 & -2.45 & 7.06 \\ 0.01 & 1.06 & 0.47 \\ 0.01 & 1.06 & 0.21 \\ 0.02 & 0.13 & 0.067 & 6.573 \\ 0.01 & 0.01 & 0.17 \\ 0.02 & 0.01 & 0.13 \\ 0.03 & 3.46 & 0.21 \\ -1.15 & -1.17 & 0.03 & 3.46 \\ 0.03 & 0.03 & 3.46 & 0.21 \\ -1.16 & 0.13 & 0.03 \\ 0.03 & 0.03 & 3.46 & 0.21 \\ -1.17 & 0.03 & 0.13 & 0.01 \\ 0.03 & 0.03 & 0.13 & 0.01 \\ 0.03 & 0.03 & 0.13 & 0.01 \\ 0.03 & 0.03 & 0.13 & 0.01 \\ 0.01 & 0.03 & 0.07 \\ 1.04 & 0.57 \\ -1.32 & 0.07 & 0.17 \\ 0.01 & 0.07 & 0.07 \\ 0.01 & 0.07 & 0.07 \\ 0.01 & 0.07 & 0.07 \\ 0.01 & 0.07 & 0.07 \\ 1.00 & 0.07 & 0.07 \\ 1.10 & -2.89 \\ 1.10 & -2.89 \\ 0.01 & 0.07 \\ 1.10 & -2.89 \\ 0.01 & 0.07 \\ 0.02 & 0.07 \\ 1.10 & -2.89 \\ 0.01 & 0.07 \\ 0.01 & 0.07 \\ 0.01 & 0.07 \\ 0.01 & 0.07 \\ 0.01 & 0.07 \\ 0.01 & 0.07 \\ 0.01 & 0.07 \\ 0.01 & 0.07 \\ 0.01 & 0.07 \\ 0.01 & 0.07 \\ 0.01 & 0.07 \\ 0.01 & 0.07 \\ 0.01 & 0.07 \\ 0.01 & 0.07 \\ 0.01 & 0.07 \\ 0.01 & 0.07 \\ 0.00 & 0.11 \\ 0.00 & 0.01 \\ 0.01 & 0.07 \\ 0.00 & 0.01 \\ 0.00 & 0.00 \\ 0.00 $	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccccc} -0.00 & 0.11 & 0.19 & -0.02 & 0.12 & 0.10 & 0.02 & 0.11 & 0.09 \\ 0.02 & 0.15 & 0.14 & 0.02 & 0.11 & 0.09 & 0.13 & 0.14 & 0.23 & 2.70 & 0.11 & 0.09 & 0.23 & -7.01 & 0.03 & 0.15 & -0.14 & -0.23 & -7.01 & 0.03 & 0.14 & -0.23 & -7.01 & 0.03 & 0.14 & -0.23 & -7.01 & 0.03 & 0.14 & -0.23 & -7.01 & 0.03 & 0.01 & 0.01 & 0.01 & 0.01 & 0.01 & 0.01 & 0.01 & 0.01 & 0.01 & 0.01 & 0.01 & 0.01 & 0.01 & 0.01 & 0.01 & 0.02 & 0.01 & 0.02 & 0.01 & 0.02 & 0.01 & 0.02 & 0.01 & 0.02 & 0.01 & 0.02 & 0.01 & 0.02 & 0.01 & 0.02 & 0.01 & 0.02 & 0.01 & 0.02 & 0.01 & 0.02 & 0.01 & 0.01 & 0.01 & 0.02 & 0.01 & 0.02 & 0.01 & 0.02 & 0.01 & 0.00 & 0.01 & 0.00 & 0.01 & 0.00 & 0.01 & 0.00$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$

Table 7: Climate risk premia

TR
and
RI
to P
ios
tfol
por
$G_E$
GH
$I_{EI},$
HG
5
ESC
щ
WU
bro
and
een
f gr
y ol
ivit
nsit
Se
e 8:
Iabl

neen	0.047444	$0.046^{***}$ 8.204	1.008***	ou.o.12 0.467***	20.907	$0.173^{***}$	5.790	-0.043	0.199***	8.970			-0.123 -0.535		reen	$0.038^{***}$	5.570	20 455	0.511***	20.274	0.293***	9.575	0.042	1.621 0.154***	5.823			$0.172 \\ 0.568$		reen	0.051***	0.080***	83.117	$0.503^{***}$	19.479	-0.012	-0./00	-1.500	$0.177^{***}$	7.745		-0.330
ڻ ت <sup>ي</sup>	0 0 0 0 0	$0.046^{**}$ 8.244	1.008***	0.467***	20.994	$0.173^{***}$	5.839	100.0-	0.199***	9.005	0.055	0.268		GE	0	0.038***	5.469	50.000	09.909 0511***	20.194	0.293 * * *	9.501	0.042	1.613	5.768	0.040	0.143		C E	G	0.053***	/.801 0.981***	82.930	$0.504^{***}$	19.735	-0.013	-0.830	-1574	$0.176^{***}$	7.744 0.284	1.014	
GHO GHO	TIMC 0 000 0 TIMC	$0.029^{***}$ 5.920	1.017***	$0.150^{***}$	6.303	$0.128^{***}$	5.317	-0.052***	-0.033	-1.326			-0.312 -1.468	CHO	uwc	$0.030^{***}$	5.259	1.00/	92.96/ 0.046**	2.041	0.011	0.477	-0.125***	-1.373	1.131			-0.076 -0.334	CHO	uwc	0.036***	4./42 1 009***	85.419	$0.243^{***}$	9.727	0.285***	0.062***	-0.00	-0.021	-0.876		-0.765**
Bro		0.029***	1.018***	0.151***	6.295	$0.127^{***}$	5.306	1 000	-0.033	-1.323	0.040	0.215			Br	$0.030^{***}$	5.090	1.00/	94.190 0.046**	2.039	0.011	0.480	-0.124***	-/.358 800.0	0.020	0.108	816.0			Bre	0.038***	4.733	86.060	$0.244^{***}$	9.711	0.283***	12.024	-0.004 -3 314	-0.022	-0.927	-0.283	
nen	0.044252	$0.044^{***}$ 7.212	1.047***	0.264***	8.935	$0.372^{***}$	10.447	-0.041*	0.092***	3.446			-0.019 -0.075		een	$0.035^{***}$	4.873	020 03	0/0720***	8.444	0.534***	16.141	-0.027	-0.892	1.373			$0.376 \\ 1.130$		een	0.053***	0.001 1 0000***	65.476	$0.416^{***}$	15.135	0.106***	6/6.C	-0.805	0.069***	3.541		-0.260 -1.010
i <sub>EI</sub> Gr	10 10 10 10 10	$0.044^{***}$ 7.211	1.047***	0.264***	8.921	0.372***	10.436	-0.041*	0.092***	3.442	0.005	0.023		JEI	G	$0.036^{***}$	4.872	021 02	0.050***	8.436	0.533***	16.091	-0.028	-0.899	1.359	-0.138	-0.517		JEI	Gr	0.056***	1 000***	64.946	$0.417^{***}$	15.100	0.105***	2.809 0.016	-0.010	0.069***	3.515 0.521*	1.902	
GHC	11M1	$0.029^{***}$ 4.991	1.024***	90.070 0.229***	8.237	$0.105^{***}$	3.465	-0.139***	0.049	1.610			-0.249 -0.985	GHG	UMU	$0.033^{***}$	4.519	1.0097	0 137***	4.373	0.005	0.148	-0.164***	-6.886	3.253			$0.112 \\ 0.387$	GHC	NWD	$0.032^{***}$	0.975***	85.273	$0.351^{***}$	14.212	$0.224^{***}$	11.952 0.121***	-6.641	0.040*	1.649		-0.921***
Brc	20000	$0.029^{***}$ 4.994	1.024***	92.960 0.230***	8.227	$0.105^{***}$	3.459 0.120***	-0.139***	0.049	1.607	-0.031	-0.145			Bro	$0.033^{***}$	4.486	1.005 CT	0.137***	4.341	0.005	0.144	-0.164***	-6.820	3.216	-0.050	-0.196			Brc	$0.033^{***}$	0.75***	86.887	$0.352^{***}$	14.119	$0.222^{***}$	0.122***	-0.136	0.039	1.567 -0.166	-0.100 -0.456	
en	0.001444	$0.021^{***}$ 7.329	$1.016^{***}$	0.082***	5.431	$0.161^{***}$	17.002	-0.043***	-0.025**	-1.997			-0.072 -0.548		ien	$0.018^{***}$	4.916		200.011	0.564	$0.160^{***}$	13.567	-0.054***	4.284	-1.496			$0.137 \\ 0.815$		en	0.029***	0.220 0 995***	110.242	$0.218^{***}$	14.417	0.145***	0.020***	-2.660	-0.002	-0.118		-0.298 -1.595
score Gre	10	$0.021^{***}$ 7.381	1.016***	0.082***	5.161	$0.161^{***}$	16.861	-0.043***	-0.025**	-1.969	$0.276^{**}$	2.488		score	Gre	$0.017^{***}$	4.584	109 011	160.011	0.550	$0.160^{***}$	13.576	-0.053***	-4.257	-1.490	0.308**	2.245		score	Gre	0.031***	0/	103.725	$0.219^{***}$	13.882	0.144*** 11 500	0.021***	-2.744	-0.002***	-0.149 0 345*	1.931	
wn ESG 5	0 0 10 40 4 4 4	$0.040^{***}$ 6.431	0.922***	90.09/ 0.398***	18.441	-0.001	-0.043	-0.111.***	-/200 0.192***	9.305			-0.087 -0.381	ESG	wn	$0.030^{***}$	4.617	0.700 007 001	0 450***	17.686	0.062***	2.761	-0.085***	-4.847	6.381			0.178 0.680	ESG	wn	0.051***	0.873***	67.457	$0.410^{***}$	15.649	-0.099***	-4.89/ 0.101***	-5.822	$0.187^{***}$	6.405		-0.313
Bro	010	$0.040^{***}$ 6.357	0.922***	0.398***	18.266	-0.001	-0.048	-0.111.***	0.192***	9.227	0.170	0.866			Bro	$0.029^{***}$	4.430	202 001	0.450***	17.683	0.063***	2.762	-0.085***	-4.825	6399	0.307	c/2.1			Bro	0.052***	0.420	67.682	$0.411^{***}$	15.574	-0.099***	-4.894 0.100***	-5 899	$0.187^{***}$	6.502 0.059	0.150	
En la	0.0004444	0.022*** 6.838	1.029***	0.101***	5.330	0.253***	18.073	-0.00/***	600.0-	-0.644			0.010 0.068		en	$0.021^{***}$	4.832	103 111	1727111	2.418	0.266***	15.110	-0.087***	-5.944	0.156			0.171 0.863		en	0.027***	270.0 %**	102.227	$0.230^{***}$	13.554	0.206***	15.814	-2.531	-0.017	-1.104		-0.160
ore Gre	2000	0.022*** 6.789	1.029***	9.101***	5.242	$0.253^{***}$	17.833	-0.00/***	-0.008	-0.629	0.145	1.202		ore	Gre	$0.021^{***}$	4.711	111 100	0.045**	2.415	0.266***	15.096	-0.087***	-5.943	0.155	0.047	0.302		ore	Gre	0.029***	0.006***	100.968	$0.231^{***}$	13.456	0.205*** 15 675	C/0.CI	-2.605	-0.017	-1.121 0 330*	1.716	
wn E sc	110	0.038*** 7.349	0.960***	0.451***	22.987	-0.010	-0.567		0.183***	9.543			-0.269 -1.353	Esc	лп	$0.029^{***}$	4.744	0.9/8/0	910.02 0.453***	18.789	0.051 **	2.376	$-0.052^{***}$	-3.018	6.322			-0.022 -0.087	Esc	vn	0.047***	0.926***	71.267	$0.527^{***}$	23.372	-0.112***	-0.441	-2.427	$0.192^{***}$	7.631		-0.463 -1 475
Brov	010	0.038*** 7.285	0.960***	0.451***	22.770	-0.010	-0.568		0.183***	9.498	0.095	0.529			Brov	$0.028^{***}$	4.594	0.9/8	90.109 0.453***	18.773	0.051**	2.376	-0.052***	-3.006	6.335	0.148	0.657			Brov	0.049***	0.026***	70.400	0.528***	23.411	-0.113***	-0.452	-0.04/	0.191***	7.532	0.484	
Full sample		t] (		MB (	t]	- IMF	t] -		MA MA			ti (	eri ti	3efore 2015		ntercept (	t]		MB	tl 1	HML	ti	- MMS			IRI	<u>ר</u>	E SKI	After 2015		ntercept (		ti ti	SMB (	<u>1</u>	- HML			MA (	ti TRI		RI

and governance score (ESG), greenhouse gas emissions intensity ( $GHG_{EI}$ ) calculated as the sum of scope 1 and 2 divided by net revenue, and greenhouse gas emissions ( $GHG_{E}$ ) calculated as the sum of scope 1 and 2. We perform time-series regressions of the brown and green portfolios excess returns on the Fama-French five factors (market, MKT; size, SMB; value, HML; profitability, RMW; and investment, CMA) plus Physical Risk Index (PRI) or Transition Risk Index (TRI) for the period Jan 2005-Oct 2021 (Full sample), Jan 2005-Dec 2014 (Before 2015), and Jan 2015-Oct 2021 (After 2015). t-statistics ([t]), and Newey-West standard are adopted. \* p < 0.1; \*\* p < 0.05; \*\*\* p < 0.01. Environmental, ESG, and GHG emissions data are sourced from Refinitiv. Note:

Table 9: Sensitivity of NACE to TRI

Full sample		B	C	D	Ш	Ľ.	U	H		   _	X		M	z	0	0	R	S
Intercent	0.061	0.034**	***600	0.044***	0.037***	0.013	0.040***	0.032***	0.018*	0.037***	0.041***	0.072**	0.041***	0.041***	* 0 00	0.068***	0.023*	-0.006
[1]	1.465	1.994	6.606	4.778	3.008	1.552	5.951	4.238	1.745	7.120	8.271	2.264	7.284	5.356	1.946	4.856	1.731	-0.450
MKT	$0.919^{***}$	$1.156^{***}$	$0.933^{***}$	0.780***	0.696***	$1.084^{***}$	$0.908^{***}$	$0.951^{***}$	$1.068^{***}$	$0.866^{***}$	$0.950^{***}$	0.747***	$0.915^{***}$	$1.055^{***}$	0.986***	0.663 ***	$0.824^{***}$	$0.827^{***}$
[t]	15.604	39.596	60.647	35.638	35.224	47.997	78.261	55.310	45.394	124.048	74.829	33.050	76.782	55.300	26.580	40.358	43.674	32.875
SMB	-0.070	$0.523^{***}$	$0.403^{***}$	0.015	0.235***	$1.055^{***}$	$0.813^{***}$	0.998***	$1.365^{***}$	$0.514^{***}$	$0.393^{***}$	$0.646^{***}$	$0.383^{***}$	$0.919^{***}$	$1.141^{***}$	$0.408^{***}$	$1.115^{***}$	0.893***
Ξ	-0.423	5.990	17.580	0.379	4.859	23.723	28.388	22.929	15.660	22.307	13.224	11.394	13.044	21.943	20.960	7.667	17.154	14.510
HML	-0.051	0.359 * * *	-0.043	-0.125***	-0.194***	$0.444^{***}$	-0.046	$0.195^{***}$	$0.349^{***}$	$-0.101^{***}$	$0.462^{***}$	$0.171^{***}$	-0.027	-0.021	-0.038	-0.239***	-0.086**	$0.321^{***}$
[1]	-0.396	4.915	-1.643	-3.296	-5.691	10.271	-1.462	5.758	6.058	-5.764	18.719	4.762	-0.989	-0.457	-0.625	-6.789	-2.266	5.779
RMW	-0.658***	$-0.170^{**}$	$-0.111^{***}$	$-0.226^{***}$	$0.166^{***}$	$0.174^{***}$	$0.106^{***}$	$0.070^{**}$	$0.431^{***}$	-0.080***	$-0.105^{***}$	$-0.126^{***}$	$-0.184^{***}$	$0.141^{***}$	0.399 * * *	-0.221***	$0.117^{***}$	$0.205^{***}$
Ξ	-4.802	-2.456	-6.958	-7.617	3.746	5.256	4.574	2.237	8.379	-4.373	-5.358	-3.781	-10.523	4.148	8.556	-5.352	2.711	4.144
CMA	$0.493^{***}$	$0.367^{***}$	$0.138^{***}$	-0.067*	-0.094*	$0.290^{***}$	$0.113^{***}$	0.013	$0.101^{*}$	0.047**	0.006	$0.289^{***}$	$0.177^{***}$	$0.147^{***}$	0.016	$0.238^{***}$	$0.197^{***}$	0.115*
[t]	3.097	3.870	6.493	-1.740	-1.860	8.617	4.459	0.341	1.719	2.055	0.211	6.836	6.035	4.686	0.282	5.434	4.568	1.792
TRI	-3.045*	0.004	0.474**	-0.589	-0.121	0.424	0.207	$0.561^{*}$	0.127	0.024	0.063	0.624	0.268	0.391	0.422	0.665	-0.369	0.119
[t]	-1.847	0.008	2.380	-1.532	-0.233	1.262	0.853	1.734	0.292	0.111	0.338	1.482	1.252	1.252	0.712	1.113	-0.643	0.207
Before 2015	A	B	C	D	Е	F	G	H	I	l	×	L L	M	z	0	0	R	s
Intercent	0.108*	0.060***	0.033***	0.056***	0.045***	0.015	0.019**	0.025***	0.008	0.037***	0.039***	0.015	0.045***	0.037***	**6600	0.072***	0.002	0.006
[1]	1 728	3 153	4 873	5 081	2.915	1 324	2.304	2.625	0.536	5 628	6 767	1 299	5 598	3 857	2 007	3 807	0 122	0.326
MKT	1 030***	1 156***	0.015***	0.732***	0.673***	1 000***	0.887***	0.800***	1 036***	0.863***	0.010***	0 774***	0.000***	1 040***	0.000***	0.670***	0.760***	0.814**
[t]	12,510	36.493	45 507	20.02	27 602	30.652	74 889	55 238	39 407	104 128	63 404	42.851	54 799	60 177	18 917	31 210	34319	22.558
SMB	-0.058	0.117	0.443***	-0000	0.338***	1 035***	0.852***	0.835***	1 170***	0 506***	0.360***	***/190	0 380***	0 970***	1 107***	0 440***	0.030***	0.814**
E	-0.234	1 325	15 736	-0.196	6 523	01 279	27 204	20.551	16 535	19 150	12,881	15 178	10.298	23.818	14 585	6 103	12,829	11 308
HMI	-0.210	0.014	0.000	-0.003*	-0.154***	0 557***	0.082***	0.000***	0.445***	-0.030*	0.517***	0.252***	0.023	0.000*	-0.052	-0.175***	-0.075	0.370***
E	1140	0.166	0.606	1 7 8 1	3 176	1000	20000	6 245	6.450	1 877	17.001	6.618	0.645	1 867	0.558	3514	0.573	2100
DMW	-1.144	0.100	0.107***	0.003***	0.131**	0.170***	0.146***	0.074**	0.472***	-1.044	0.158***	0.107***	0.186***	0.173***	0.307***	0.018***	0800	0.018***
A ININ		-2.443	-4 702	-0.2.02	2 196	3.576	4 912	2.054	8 670	-0.122	-0.136	-4.710	-7.786	C(110	6.243	-0.210	0.000	2 004
CMA	0.563***	0.707***	0 133***	-0.00*	150	0.250***	0.073**	0.057	0.110*	-0.00-	0.043	0.006***	0.181***	0.164***	0.174 ***	0 173 ***	0.748***	0175**
[1]	2.615	8.265	4.663	-1.942	-2.156	6.303	2.237	1.509	1.766	-0.738	1.326	4.922	4.608	5.397	2.842	3.047	4.821	2.111
TRI	-4.337*	-0.056	$0.841^{***}$	-0.900*	-0.405	0.341	0.468	0.827 * *	0.484	0.029	-0.099	0.760	0.298	0.498	0.993	1.017	-0.223	-0.129
Ξ	-1.951	-0.091	3.121	-2.032	-0.657	0.793	1.567	2.256	0.883	0.116	-0.427	1.516	0.990	1.281	1.260	1.288	-0.312	-0.172
After 2015		B	C		    ш	Ц	0	H		-	×	[    _1	X	z	0	0	R	S
Intercept	0.018	0.017	$0.023^{***}$	0.027*	0.024	0.007	$0.064^{***}$	$0.041^{***}$	0.028	$0.032^{***}$	$0.039^{***}$	0.027*	$0.032^{***}$	$0.042^{***}$	0.004	0.051***	$0.055^{**}$	-0.020
[1]	0.510	0.662	3.691	1.659	1.319	0.547	6.961	2.866	1.603	4.026	6.767	1.831	4.543	3.260	0.205	2.750	2.562	-0.992
MKT	0.754***	$1.136^{***}$	0.958***	0.853***	0.739***	$1.063^{***}$	$0.934^{***}$	$1.016^{***}$	$1.094^{***}$	0.873***	$0.912^{***}$	***6/1/0	0.920 * * *	$1.069^{***}$	1.079 * * *	$0.652^{***}$	0.899***	$0.833^{***}$
Ξ	13.133	33.200	73.758	29.179	21.313	49.199	64.873	42.316	28.970	69.284	63.404	19.525	66.340	31.827	35.161	28.689	33.336	29.325
SMB	0.044	$0.753^{***}$	$0.379^{***}$	-0.001	0.080	$1.246^{***}$	$0.858^{***}$	$1.211^{***}$	$1.737^{***}$	$0.469^{***}$	$0.360^{***}$	$0.687^{***}$	$0.423^{***}$	$0.926^{***}$	$1.004^{***}$	$0.451^{***}$	$1.408^{***}$	$1.055^{***}$
Ξ	0.278	7.908	12.589	-0.013	0.780	19.967	22.017	20.215	12.750	15.722	12.881	6.550	11.930	11.638	14.552	5.597	14.069	13.782
HML	-0.049	$0.714^{***}$	-0.134***	-0.085*	-0.160***	$0.171^{***}$	-0.245***	$0.143^{***}$	0.101	-0.163***	$0.517^{***}$	0.075	-0.135***	-0.236***	0.018	-0.344***	-0.222***	$0.143^{**}$
[1]	-0.308	10.171	-5.915	-1.689	-3.031	3.719	-10.107	2.885	1.538	-6.900	17.021	1.579	-5.071	-5.315	0.296	-6.672	-3.492	2.451
RMW	-0.200	-0.336***	-0.103***	-0.232***	$0.260^{***}$	$0.164^{***}$	$0.071^{**}$	0.053	$0.344^{***}$	0.012	-0.158	0.010	-0.185***	0.093*	$0.396^{***}$	-0.213***	$0.159^{**}$	$0.150^{***}$
Ξ	-1.494	-4.639	-4.540	-4.845	4.359	4.183	2.527	1.084	4.087	0.449	-6.321	0.190	-7.080	1.941	6.529	-3.557	2.502	2.610
CMA	0.223	0.129	0.097***	0.064	0.005	$0.156^{***}$	0.058*	0.004	0.006	0.096***	0.043***	$0.335^{***}$	$0.104^{***}$	-0.021	-0.129	0.258***	0.130*	-0.062
Ξ	1.478	1.489	3.633	0.967	0.070	3.139	1.799	0.070	0.064	2.946	1.326	4.903	3.937	-0.432	-1.528	3.760	1.742	-0.777
TRI	-2.601	-0.553	-0.308	0.006	0.384	0.284	0.236	0.292	-0.449	-0.038	-0.099	0.464	0.014	0.227	-0.753	-0.311	-0.084	0.210
Ξ	-1.503	-0.499	-1.110	0.008	0.422	0.530	0.577	0.421	-0.575	-0.101	-0.427	0.672	0.050	0.422	-0.915	-0.396	-0.086	0.242
ote: We gr	oup the	EuroStox	x 600 In	dex histo	rical con	Istituents	into NA	CE secto	rs: Agri	culture.	forestry	and fishin	ng (A). N	<b>dining</b> an	d quarry	ving (B).	Manufac	turing (C).
	and days				ĺ				0	( )	(			n (	frank a			
lectricity, g	as, stean	n and air	condition	idns guiu	JV (U),	Water su	pply; sev	verage, v	waste ma	unagemet	it and re	mediation	n activitie	3S (E), C	onstruction	on (F), V	Vholesale	and retail

Financial and insurance activities (K), Real estate activities (L), Professional, scientific and technical activities (M), Administrative and support service activities (N), Public trade; repair of motor vehicles and motorcycles (G), Transportation and storage (H), Accommodation and food service activities (I), Information and communication (J), administration and defence; compulsory social security (O), Human health and social work activities (Q), Arts, entertainment and recreation (R), Other service activities (S). We perform time-series regressions of the NACE sectors excess returns on the Fama-French five factors (the market factor, MKT; the size factor, SMB; the value factor, HML; the profitability factor, RMW; and the investment factor, CMA) plus Transition Risk Index (TRI) over the period Jan 2005-Oct 2021 (Full sample), Jan 2005-Dec 2014 (Before 2015), and Jan 2015-Oct 2021 (After 2015). t-statistics ([t]), and Newey-West standard are adopted. \* p < 0.1; \*\* p < 0.05; \*\*\* p < 0.01. ž Ĕ

Table 10: Sensitivity of NACE to PRI

Full sample	A	в	C	D	н	н	G	H		   -	×	Г	M	z	0	0	R	S
Intercept	0.061	$0.034^{**}$	$0.033^{***}$	$0.044^{***}$	$0.037^{***}$	0.013	$0.040^{***}$	$0.032^{***}$	0.018*	$0.037^{***}$	$0.041^{***}$	0.022**	$0.041^{***}$	$0.041^{***}$	0.022*	0.068***	0.023*	-0,006
[t]	1.466	1.992	6.581	4.750	3.037	1.554	5.976	4.242	1.746	7.132	8.319	2.270	7.289	5.369	1.947	4.876	1.728	-0.450
MKT	$0.920^{***}$	$1.155^{***}$	$0.933^{***}$	$0.780^{***}$	$0.697^{***}$	$1.083^{***}$	0.908***	$0.951^{***}$	$1.068^{***}$	0.865***	0.950***	$0.746^{***}$	$0.915^{***}$	$1.054^{***}$	0.987***	$0.663^{***}$	$0.824^{***}$	$0.826^{***}$
Ξ	15.741	39.588	60.500	35.631	34.818	47.971	77.954	55.223	45.335	123.484	75.016	33.020	76.784	55.286	26.585	40.543	43.658	32.995
SMB	-0.068	$0.522^{***}$	$0.402^{***}$	0.014	$0.235^{***}$	$1.054^{***}$	$0.813^{***}$	$0.997^{***}$	$1.365^{***}$	$0.513^{***}$	$0.394^{***}$	$0.645^{***}$	$0.383^{***}$	$0.918^{***}$	$1.142^{***}$	$0.407^{***}$	$1.116^{***}$	$0.891^{***}$
Ξ	-0.410	5.994	17.480	0.353	4.825	23.717	28.412	22.885	15.640	22.288	13.346	11.357	13.034	21.867	20.967	7.588	17.115	14.580
HML	-0.051	$0.359^{***}$	-0.043	-0.125***	-0.194***	$0.444^{***}$	-0.046	$0.195^{***}$	$0.349^{***}$	$-0.101^{***}$	$0.462^{***}$	$0.171^{***}$	-0.027	-0.021	-0.038	-0.239***	-0.086**	$0.321^{***}$
[t]	-0.392	4.918	-1.640	-3.285	-5.621	10.263	-1.459	5.749	6.059	-5.748	18.915	4.755	-0.991	-0.457	-0.628	-6.837	-2.267	5.815
RMW	-0.659***	-0.170 * *	$-0.111^{***}$	-0.226***	$0.166^{***}$	$0.174^{***}$	$0.106^{***}$	$0.070^{**}$	$0.431^{***}$	-0.079***	-0.105***	$-0.126^{***}$	$-0.184^{***}$	$0.141^{***}$	0.399***	-0.221 * * *	$0.117^{***}$	$0.206^{***}$
[t]	-4.825	-2.454	-6.973	-7.621	3.721	5.254	4.580	2.245	8.373	-4.372	-5.390	-3.769	-10.512	4.153	8.548	-5.399	2.698	4.149
CMA	$0.497^{***}$	$0.367^{***}$	$0.137^{***}$	-0.066*	-0.094*	$0.290^{***}$	$0.113^{***}$	0.012	0.100*	0.047**	0.005	$0.288^{***}$	$0.177^{***}$	$0.146^{***}$	0.016	$0.237^{***}$	$0.198^{***}$	$0.115^{*}$
Ξ	3 148	3 871	6 443	-1 708	-1859	8 584	4 445	0 321	1 717	2 063	0.208	6 798	6.032	4 671	0.770	5 300	4 586	1 802
DR1	-0.981	-0.748	-0.107	-0.035**	0.217	0.181	-0.164	-0.184	0.081	-0.550**	0.326*	0 102	0.145	0.071	0.483	0.173	0 100	-0.633
Ξ	-0.564	-0.366	-0.506	-2.260	0.400	0.482	-0.585	-0.107	0.152	-2.421	1.708	0.244	0.596	0.198	0.846	0.280	0.165	-0.954
Before 2015	A	B	C		Ш	Ц	0	H		ſ	X	L L	M	z	0	0	L N	S
Intercent	0.04	0.050***	0.036***	0.055***	0.044***	0.016	0.001**	***LCU U	0.010	0.038***	0.038***	0.017	0.045***	0.038***	0.031 **	0.074***	0.001	0.006
	1 575	2 100	LOC 2	0.000	102.0	1 200	2 500	2005	0100	5 766	6.691	100	C112 S	2 000	2 1 95	3 800	100.0	0.362
[4] MIVT	1.020***	J.107 1 157***	0.015***	4.040	0.672***	1.000***	COC.7	***0000	1.026***	0.07.00	0.001	**** <b>5000</b>	0.000***	1 0.11***	***0000	0.671 ***	***0920	0.000
	002 01	L/370	C16.0	761.0	011 20	060.1	100.0	54004	0001	C00.0	1.216.0	10,400	202.02	1.041	0.020.01	10.00	00/.0	00200
Ē	660.21	140.05	40.192	c0/.67	21.119	0/0.06	/4.052	04.994	59.334	104.05	03.885	42.482	04./80	01110	18.928	186.06	34.302	07 5.77
SMB	10.0-	0.119	0.441***	-0.010	0.339***	1.035***	0.851***	0.835***	1.17/***	0.095***	$0.362^{***}$	0.6/8***	0.389***	0.9/0***	1.128***	0.441***	0.930***	0.813***
Ξ	-0.208	1.336	15.511	-0.208	6.469	21.259	27.032	20.246	16.577	19.123	12.892	14.957	10.301	24.065	14.610	6.139	12.808	11.268
HML	-0.215	0.014	0.021	-0.092*	-0.153***	$0.556^{***}$	$0.082^{**}$	$0.219^{***}$	$0.445^{***}$	-0.039*	$0.517^{***}$	$0.252^{***}$	0.022	0.089*	-0.053	-0.177 * * *	-0.025	$0.373^{***}$
Ē	-1.134	0.165	0.582	-1.762	-3.115	8.784	2.523	6.138	6.438	-1.821	17.163	6.547	0.635	1.822	-0.572	-3.508	-0.521	4.493
RMW	-1.032***	$-0.161^{**}$	$-0.108^{***}$	$-0.201^{***}$	$0.131^{**}$	$0.169^{***}$	$0.145^{***}$	0.073**	$0.472^{***}$	$-0.121^{***}$	$-0.158^{***}$	-0.198 * * *	$-0.187^{***}$	$0.172^{***}$	$0.396^{***}$	-0.219***	0.080	$0.218^{***}$
[t]	-5.231	-2.449	-4.758	-5.712	2.177	3.521	4.890	2.021	8.668	-5.029	-6.336	-4.754	-7.807	4.196	6.224	-3.914	1.436	2.999
CMA	$0.567^{***}$	$0.727^{***}$	$0.131^{***}$	+060.0-	$-0.143^{**}$	$0.259^{***}$	$0.073^{**}$	0.055	0.109*	-0.022	0.043	$0.224^{***}$	$0.181^{***}$	$0.164^{***}$	$0.173^{***}$	$0.171^{***}$	$0.248^{***}$	$0.176^{**}$
[t]	2.665	8.283	4.601	-1.891	-2.129	6.276	2.208	1.455	1.764	-0.733	1.330	4.847	4.616	5.347	2.812	3.040	4.824	2.121
PRI	-0.452	0.711	0.010	-0.946*	0.064	0.241	0.010	0.442	-0.345	-0.408	0.475**	0.686	0.441	0.511	1.094	0.994	0.161	-0.623
[t]	-0.176	0.968	0.036	-1.882	0.096	0.489	0.028	0.981	-0.500	-1.453	2.071	1.355	1.343	1.124	1.416	1.163	0.223	-0.735
After 2015	A	В	С	D	Е	Н	ŋ	Н	_	J	K	L	М	z	0	0	R	S
Intercept	0.026	0.014	$0.024^{***}$	0.024	0.023	0.006	$0.062^{***}$	$0.037^{***}$	0.032*	$0.031^{***}$	$0.046^{***}$	0.024	$0.031^{***}$	$0.039^{***}$	0,006	$0.050^{***}$	$0.057^{***}$	-0.024
[t]	0.722	0.571	4.023	1.539	1.309	0.479	7.075	2.709	1.954	4.021	5.724	1.625	4.620	3.288	0.334	2.704	2.783	-1.187
MKT	$0.755^{***}$	$1.136^{***}$	$0.958^{***}$	$0.852^{***}$	$0.739^{***}$	$1.063^{***}$	$0.933^{***}$	$1.015^{***}$	1.095 * * *	$0.873^{***}$	***666.0	$0.778^{***}$	$0.920^{***}$	$1.068^{***}$	$1.080^{***}$	$0.651^{***}$	0.899***	$0.833^{***}$
Ξ	13.344	33.262	74.030	28.942	21.414	49.206	64.928	42.411	28.881	68.577	79.506	19.546	66.847	31.856	35.489	28.781	33.132	29.225
SMB	0.042	$0.749^{***}$	$0.379^{***}$	-0.002	0.081	$1.246^{***}$	$0.857^{***}$	1.209 * * *	$1.738^{***}$	$0.468^{***}$	0.464***	$0.686^{***}$	$0.422^{***}$	$0.925^{***}$	1.003 * * *	$0.449^{***}$	$1.408^{***}$	$1.053^{***}$
[t]	0.262	7.902	12.527	-0.039	0.782	19.910	22.024	20.210	12.794	15.726	12.040	6.474	11.840	11.528	14.835	5.550	14.059	13.529
HML	-0.047	$0.714^{***}$	$-0.133^{***}$	-0.085*	$-0.160^{***}$	$0.171^{***}$	-0.245***	$0.142^{***}$	0.101	$-0.163^{***}$	$0.360^{***}$	0.075	-0.135***	-0.236***	0.018	-0.344***	-0.222***	$0.143^{**}$
[t]	-0.305	10.233	-5.927	-1.681	-3.033	3.716	-10.127	2.886	1.547	-6.903	13.536	1.565	-5.089	-5.341	0.303	-6.732	-3.494	2.409
RMW	-0.205	-0.338***	$-0.104^{***}$	-0.233***	$0.261^{***}$	$0.164^{***}$	0.072**	0.053	$0.344^{***}$	0.012	-0.028	0.010	-0.185***	0.093 * =	$0.394^{***}$	-0.214***	$0.159^{**}$	$0.150^{***}$
Ξ	-1.540	-4.669	-4.566	-4.857	4.367	4.185	2.529	1.083	4.079	0.428	-1.126	0.195	-7.105	1.945	6.454	-3.572	2.497	2.598
CMA	0.228	0.130	$0.098^{***}$	0.064	0.005	$0.155^{***}$	0.057*	0.003	0.007	$0.096^{***}$	-0.081***	$0.334^{***}$	$0.104^{***}$	-0.022	-0.128	0.259***	$0.130^{*}$	-0.063
[1]	1.497	1.495	3.657	0.958	0.061	3.120	1.786	0.057	0.076	2.928	-2.597	4.861	3.929	-0.446	-1.499	3.769	1.743	-0.766
PRI	-2.184	-2.420**	-0.299	-1.006	0.464	0.065	-0.122	-1.058*	0.826	-0.697**	0.452	-0.574	-0.364	-0.564	-0.615	-1.124	0.361	-0.868
[1]	-1.151	-2.392	-0.963	-1.438	0.574	0.123	-0.289	-1.828	066.0	-2.012	1.345	-0.841	-1.090	-1.008	-0.698	-1.308	0.358	-0.840
ote. We or	oun the	FurnStoy	v 600 In	dev histo	rical cor	stiments	into NA	CF secto	ne. A ori	- on the	forestry :	and fichin	M (A) N	lining an	d anarry	ing (B)	Manufac	turing (C)
	vin quo		· · · ·				· · · · · · · · · · · · · · · · · · ·		1917 .010	, (111111)			יי יוריא פר	mne n	frmnh ni	me (v);		
lectricity, g	as, stear	n and air	conditio	ldns guiu	oly (D),	Water su	pply; sev	verage, v	vaste ma	unagemer	it and rei	mediation	n activitie	s (E), C	onstruction	on (F), V	Vholesale	and retail

Financial and insurance activities (K), Real estate activities (L), Professional, scientific and technical activities (M), Administrative and support service activities (N), Public trade; repair of motor vehicles and motorcycles (G), Transportation and storage (H), Accommodation and food service activities (I), Information and communication (J), administration and defence; compulsory social security (O), Human health and social work activities (Q), Arts, entertainment and recreation (R), Other service activities (S). Ĕ ž

We perform time-series regressions of the NACE sectors excess returns on the Fama-French five factors (the market factor, MKT; the size factor, SMB; the value factor, HML;

the profitability factor, RMW; and the investment factor, CMA) plus Physical Risk Index (PRI) over the period Jan 2005-Oct 2021 (Full sample), Jan 2005-Dec 2014 (Before

2015), and Jan 2015-Oct 2021 (After 2015). t-statistics ([t]), and Newey-West standard are adopted. \* p < 0.1; \*\* p < 0.05; \*\*\* p < 0.01.



Figure 1: Word clouds summary for physical and transition risk vocabularies

(a) Physical risk vocabulary

(b) Transition risk vocabulary

Note: Word cloud summaries for the physical risk (a) and transition risk (b) vocabularies. Term sizes depend on the relative importance of the term according to the individual term frequency-inverse document frequency, tf-idf, score. Reported terms are the reconstructed stemmed terms. Major acronyms: Representative Concentration Pathways (RCP), hydrofluorocarbon (HFC), hydrochlorofluorocarbon (HCFC). Table A2 reports the full list of acronyms.





Figure 2: Physical and transition risk concern timeseries 2005-2021

#### Figure 3: Distribution of exposure metrics



Note: Environmental score (a), ESG score (b), GHG emissions (c), and GHG emissions intensity (d) boxplots for the NACE Rev. 2 sectors: Agriculture, forestry and fishing (A); Mining and quarrying (B); Manufacturing (C); Electricity, gas, steam and air conditioning supply (D); Water supply, sewerage, waste management and remediation activities (E); Construction (F); Wholesale and retail trade, repair of motor vehicles and motorcycles (H); Accommodation and food service activities (I); Information and communication (J); Financial and insurance activities (K); Real estate activities (K); Professional, scientific and technical activities (M); Administrative and support service activities (N); Public administration and defence, compulsory social security (O); Human health and social work activities (Q); Arts, enterteinment and recreation (R); Other services activities (S). Environmental, ESG, and GHG emissions data are sourced from Refinitiv.



Figure 4: Cumulative performances of portfolios sorted on TRI and PRI

Note: Cumulative daily performances of the low-minus-high (LMH) quintiles (light-grey lines), deciles (grey lines) and 25 percentiles (black lines) transition (a) and physical (b) climate beta portfolios considering EuroStoxx 600 Index historical constituents stocks over the period Jan 2005-Oct 2021.



Figure 5: Stock market index sensitivity to climate risks by country pre- and post-2015

(**b**) Countries sensitivity to physical climate risk ( $\beta^{PRI}$ )

Note: Stock market index returns beta sensitivity to the Transition Risk Index, TRI (a) and Physical Risk Index, PRI (b) for Austria, FTSE Austria (FTWIAUTE); Belgium, BEL 20 (BFX); Bosnia-Herzegovina, BIRS (BIRS1); Bulgaria, BSE SOFIX (SOFIX); Croatia, CROBEX (CRBEX); Cyprus, Cyprus Main Market (CYMAIN); Czech Republic, PX (PX); Denmark, OMX Copenhagen 20 (OMXC20); Estonia, Tallinn SE General (OMXTGI); Finland, OMX Helsinki 25 (OMXH25); France, CAC 40 (FCHI); Germany, DAX (GDAXI); Greece, Athens General Composite (ATG); Hungary, Budapest SE (BUX); Iceland, ICEX Main (OMXIPI); Ireland, ISEQ Overall (ISEQ); Italy, FTSE MIB (FTMIB); Latvia, Riga General (OMXRGI); Lithuania, Vilnius SE General (OMXVGI); Malta, MSE; Netherlands, AEX (AEX); Norway, Oslo OBX (OBX); Poland, WIG20 (WIG20); Portugal, PSI (PSI20); Romania, BET (BETI); Russia, MOEX Russia (IMOEX); Serbia, Belex 15 (BELEX15); Slovakia, SAX (SAX); Slovenia, Blue-Chip SBITOP (SBITOP); Spain, IBEX 35 (IBEX); Sweden, OMX Stockholm 30 (OMXS30); Switzerland, SMI (SSMI); Turkey, BIST 100 (XU100); Ukraine, PFTS (PFTSI); United Kingdom, FTSE 100 (FTSE). Pre- and post-2015 sample periods are analysed for each country. \*\*\*, \*\*, \* indicate p-value< 0.01, p-value< 0.05, and p-value< 0.10, respectively.



### Figure 6: Stock market index sensitivity to climate risks map by country pre- and post-2015

(**b**) Countries sensitivity to TRI post-2015



Note: Countries' stock market sensitivity to transition risk, TRI, pre-2015 (a) and post-2015 (b). Countries' stock market sensitivity to physical risk, PRI, pre-2015 (c) and post-2015 (d). See the stock market indices list from Figure 5.

# Appendix

Year	Source	Title	Transition	Physical
1990	IPCC	IPCC Synthesis Report 1990	115-148p	-
1990	IPCC	Climate change: The IPCC Impacts Assessment	-	Entire
1992	IPCC	Climate change: The IPCC 1990 and 1992 Assessments		87-113p
1999	IPCC	IPCC Special Report: Aviation and the global atmosphere	Entire	
2000	IPCC	IPCC Special Report: Methodological and technological issues in technology transfer	Entire	
2001	IPCC	IPCC Synthesis Report 2001	302-354p	
2001	IPCC	Climate change 2001: Impacts, adaptation and vulnerability		Entire
2005	IPCC	IPCC Special Report: Carbon dioxide capture and storage	Entire	
		IPCC Special Report: Safeguarding the ozone layer and the		
2005	IPCC	global climate system: Issues related to hydrofluorocarbons	Entire	
		and perfluorocarbons		
2007	IPCC	IPCC Synthesis Report 2007	55-70p	
2007	IPCC	Climate change 2007: Impacts, Adaptation and Vulnerability		Entire
2011	IPCC	IPCC Special Report: Renewable energy sources and climate change mitigation	Entire	
2012	IPCC	IPCC Special Report: Managing the risks of extreme events and disasters to advance climate change adaptation		Ch. 2-4
2014	IPCC	IPCC Synthesis Report 2014	75-112p	
2014	IPCC	Climate change 2014: Impacts, adaptation and vulnerability		Part A & B
2018	UNEP FI	Navigating a new climate. Part 2: Physical risks and opportunities		Entire
2019	IPCC	IPCC Special Report: Global warming of 1.5C	Ch. 2 & 4	Ch. 3
2019	IPCC	IPCC Special Report: Climate change and land		Ch. 1-5
2019	IPCC	IPCC Special Report: The ocean and cryosphere in a changing climate		Entire
2020	IMF JM	The effects of weather shocks on economic activity: What are the channels of impact?		Entire
2020	MGI	Climate risk and response: Physical hazards and socioeconomic impacts		Entire
2020	SRI	Natural catastrophes in times of economic accumulation and climate change		Entire

Table A1: List of climate change white papers for transition and physical risk

Note: This table reports the year of publication, source, title of the list of texts used to construct the physical and transition risk vocabularies. List of acronyms: IPCC, Intergovernmental Panel on Climate Change; IMF JM, International Monetary Fund - Journal of Macroeconomics; UNEP FI, United Nations Environment Programme Finance Initiative; MGI, McKinsey Global Institute; SRI, Swiss Re Institute.

Physical risk vocabulary acronyms			
GHG	Greenhouse gas		
RCP	RCP Representative Concentration Pathway		
IPCC	Intergovernmental Panel on Climate Change		
Transition risk vocabulary acronyms			
EJ/yr	Exajoules per-year		
MtCO2	Megatonne of carbon		
eq/yr	Equivalent per-year		
MtCO2 eq	Megatonne of carbon equivalent		
GHG	Greenhouse gas		
TCO2	Tonne of carbon		
GtCO2	Gigatonne of carbon		
TEAP	Technology and Economic Assessment Panel		
HCF	Hydrofluorocarbon		
TWh/yr	Terawatt hours/year		
HCFC	Hydrochlorofluorocarbon		
UNEP	United Nations Environment Programme		
IPCC	Intergovernmental Panel on Climate Change		
UNFCCC	United Nations Framework Convention on Climate Change		
IEA	International Energy Agency		
USD/kWh	United States Dollar/Kilowatt hour		

Table A2: Physical risk and transition risk vocabularies list of acronyms

Note: Physical risk and transition risk summary vocabularies as in figure 1 list of acronyms.

#### A2. Common Metrics for Firms' Climate Risk Exposure: A Review

#### A2.1. Transition Risk

Academics, practitioners, and supervisors typically use GHG emissions or GHG emissions intensity (GHG emissions scaled by some organization-specific metric) to proxy a firm's exposure to transition risk, motivated by the fact that carbon-intensive activities are likely affected by GHG emissions reduction policies (Ardia et al. 2023; Bolton and Kacperczyk 2023; In et al. 2019; NGFS 2020). However, empirical findings based on these measures are not conclusive. Bolton and Kacperczyk (2023) provide evidence of the existence of a carbon premium while considering both emissions levels and changes, but no relation with carbon intensity exists. Also Bolton and Kacperczyk (2021) and Ramelli et al. (2021) find that transition risks related to carbon emissions are priced. In contrast, In et al. (2019) find that green firms

outperform brown firms when considering carbon intensity and Hsu et al. (2023) show that a long-short portfolio constructed from firms with high versus low emission intensity generates a positive excess return.

Other potential measures of climate exposure are E and ESG scores, which aim to measure the environmental, or environmental, social and governance-related performance of a company.<sup>25</sup> A number of academic studies rely on E/ESG scores (possibly in combination with other company-specific metrics) to identify climate-sensitive companies. Görgen et al. (2020), for example, build a "greenness" score based on carbon intensity, ESG scores and an adaptability score. Alessi et al. (2021) combines ESG disclosure scores with quantitative measures on emissions, while Engle et al. (2020) focus exclusively on the E score.

Other studies consider sectoral classifications and define climate-sensitive companies as those belonging to high GHG emissions sectors. This approach is particularly relevant in contexts where the lack of transparent indicators (e.g. ESG ratings) may limit the ability of investors to understand to steer their investment toward climate-hedged portfolios (Bolton and Kacperczyk 2021). As such, Choi et al. (2020) finds that the sectors identified as major emitters by the Intergovernmental Panel on Climate Change (IPCC) earn lower stock returns than other firms. Bolton and Kacperczyk (2023) argue that relying on a sectoral analysis may facilitate the detection of climate-risky investments, whereas such an approach may ignore relevant intra-sectoral differences. Batten et al. (2016) document the impact of transition risk on the energy sector and concludes that only renewable energy companies generate abnormal returns.

#### A2.2. Physical Risk

Measuring firms' exposure to physical risk is challenging as physical risk arises from the interaction of hazard (occurrence, or probability of occurrence, of a physical event), exposure (presence of elements in areas and settings that could be adversely affected), and vulnerability (predisposition of exposed elements to suffer damages due to the hazardous event). Such dimensions typically depend on both

<sup>&</sup>lt;sup>25</sup>The exact information contained in any ESG score depends on the methodology used to calculate it, which, in turn, differs across credit/rating providers. Existing research documents large differences between ESG ratings (Chatterji et al. 2016; Gibson Brandon et al. 2021) and elaborates on the possible reasons for these Berg et al. (2022).

local/specific and macro factors. Currently, most of the information on physical risk exposure is provided by some public sources (e.g. EC JRC Risk Data Hub) and private data providers. These databases, however, are not fully comparable as they focus on different risk aspects, types of hazard and types of entities. Due to data limitations, studies that explore the consequences of physical risks on asset prices mainly focus on specific physical events and/or consider only some dimensions of physical risk (see Addoum et al. (2020); Hong et al. (2019); Kruttli et al. (2021)). Alternatively, as is the case for transition risk, sectors can be used to proxy physical risk exposure. While all sectors can suffer from natural disasters, some sectors, including energy, transportation, and telecommunications, are expected to be more exposed to climate hazards through their infrastructure assets (ECB 2021). Primary economic activities, including agriculture, forestry, fishing, mining and quarrying, are exposed through the natural and/or food systems on which they depend. Among services, the insurance sector, tourism and health care might be particularly sensitive to physical risk (IPCC 2014).

While E, ESG, and GHG emissions have mainly been used to capture exposure to transition, rather than physical risk, this distinction is not always clear and these metrics' potential to capture physical exposure has been largely unexplored. In this study, we decided to test all of the above-mentioned exposure metrics in light of their potential use for investors to hedge against physical and/or transition risks.